#### **RESEARCH ARTICLE**



# Optimizing design and dispatch of a renewable energy system with combined heat and power

Jusse Hirwa<sup>1</sup>  $\cdot$  Oluwaseun Ogunmodede<sup>1</sup>  $\cdot$  Alexander Zolan<sup>2</sup>  $\cdot$  Alexandra M. Newman<sup>1</sup>

Received: 6 December 2020 / Revised: 7 August 2021 / Accepted: 10 August 2021 / Published online: 9 January 2022 © The Author(s), under exclusive licence to Springer Science+Business Media, LLC, part of Springer Nature 2021

# Abstract

We embellish a mixed-integer program that prescribes a set of renewable energy, conventional generation, and storage technologies to procure, along with a corresponding dispatch strategy. Specifically, we add combined heat and power to this set. The model minimizes fixed and operational costs less incentives for the use of various technologies, subject to a series of component interoperability and system-wide constraints. The resulting mixed-integer linear program contains hundreds of thousands of variables and constraints. We demonstrate how to efficiently formulate and solve the corresponding instances such that we produce near-optimal solutions in minutes. A previous rendition of the model required hours of solution time for the same instances.

Keywords Mixed-integer linear optimization  $\cdot$  Renewable energy  $\cdot$  Combined heat and power systems  $\cdot$  Efficient formulations

# **1** Introduction

Distributed generation is gaining increasing interest in the energy sector owing to its economic, technical, and environmental benefits. As opposed to purchasing power exclusively from the grid, users can invest in on-site generation using technologies of their choice, such as wind, solar photovoltaics (PV) and storage. When integrated with combined heat and power (CHP) technology, a single energy source can simultaneously generate electricity and heat to meet heating and cooling demands (see Fig. 1). In other words, the process of generating electricity releases waste heat which CHP can

Alexandra M. Newman anewman@mines.edu

<sup>&</sup>lt;sup>1</sup> Colorado School of Mines, 1500 Illinois St., Golden, CO 80401, USA

<sup>&</sup>lt;sup>2</sup> National Renewable Energy Laboratory, Thermal Energy Systems Group, 15013 Denver West Parkway, Golden, CO 80401, USA



**Fig. 1** A notional distributed generation system with a collection of technologies (including thermal energy storage (TES)) available for electrical, heating, and cooling loads; the technologies highlighted in blue represent a baseline case without distributed resources. Dashed boxes on the right side of the image represent loads that depend on cooling dispatch decisions when an absorption chiller is available. Image adapted from: Anderson et al. (2021)

capture to produce usable thermal energy, offsetting the consumption of extra fuel for this purpose. In this way, distributed generation systems achieve greater energy efficiency relative to that of conventional generators that separate electrical and thermal production (Kerr 2008). The use of renewable technologies and the efficiency gains of CHP lead to significant reduction in emissions, which promote the world's initiative to reduce global pollution and meet climate change goals. Additionally, research shows that distributed generation systems offer energy savings and play a major role in reducing investments in transmission and distribution capacity (El-Khattam and Salama 2004; Gumerman et al. 2003). Benefits also include peak shaving, as well as improved system reliability and resiliency (Chiradeja and Ramakumar 2004). Our research informs the optimal design (i.e., size and mix) and dispatch of renewable technologies with combined heat and power to reduce costs for representative commercial buildings.

The National Renewable Energy Laboratory has developed REopt Lite, a model that helps energy planners assess the economic feasibility of using renewable energy technologies, combined heat and power, conventional generators, and storage (Anderson et al. 2021; Mishra et al. 2021). This model determines the system sizes and dispatch decisions, includes an option to assess grid resilience in case of an outage, and incorporates sophisticated pricing structures. For the purpose of this paper, we refer to this model as the original formulation ( $\overline{R}$ ). Ogunmodede et al. (2021) improve the performance of the mathematical formulation without CHP, but omit implementation details. Our model, which we term ( $\widehat{R}$ ), extends the reformulation in Ogunmodede et al. (2021) to include the option of CHP technologies and thermal energy storage. This involves the addition of: (i) fuel constraints, (ii) thermal production restrictions, (iii) storage operations, (iv) charging rates, (v) cold and hot thermal loads, (vi) load balancing and grid sales, and (vii) standby charges. We correspondingly provide implementation details.

In aggregate, the contributions of our paper are as follows: (i) the extension of an existing energy design and dispatch model (Ogunmodede et al. 2021) to accommodate combined heat and power technologies, (ii) an improvement in the tractability of this (and the Ogunmodede et al. (2021)) model through the use of appropriate data handling and data structures, and thorough reformulation; and, (iii) the presentation of managerial insights gained from solutions to realistic instances of this complicated system. The remainder of the paper is organized as follows: Section 2 reviews the relevant literature. Section 3 presents the notation and corresponding mathematical formulation. Section 4 provides the solution methodology we employ to increase tractability of the model. Section 5 describes the data we use and corresponding results, including performance characteristics and solution analysis. Finally, Section 6 concludes and proposes future work.

# 2 Literature review

Models that optimally determine design and dispatch simultaneously are NP-hard, and can consist of nonlinear functional forms (De Mel et al. 2020; Pruitt et al. 2014; Zakrzewski 2017) and/or integrality restrictions on (some of) the decision variables (Merkel et al. 2015). Problem simplifications, such as shortening the time horizon (Fuentes-Cortés and Flores-Tlacuahuac 2018; Gopalakrishnan and Kosanovic 2014, 2015), aggregating time periods (Oluleye et al. 2018), or scaling down the entire system (Adam et al. 2015; Merkel et al. 2015) might compromise the quality of the solution, even if the model itself becomes more tractable.

An increasing number of models in literature simultaneously address the design and dispatch problem. Specifically, there have been those that consider dispatching microgrids (Goodall et al. 2019; Scioletti et al. 2017; Zhao et al. 2014), concentrated solar power (Hamilton et al. 2020a, b), and oxidized fuel cells (Anyenya et al. 2018). Some optimization models incorporate combined heat and power. For example, Krug et al. (2020) provide a nonlinear model whose solution dispatches district heating networks, and Rong and Lahdelma (2007) weigh the cost of investing in such technologies against CO<sub>2</sub> emissions in a multi-period stochastic optimization model. Literature demonstrates alternative solutions to design and dispatch with CHP using multi-objective optimization (Hollermann et al. 2020; Huster et al. 2019; Perera et al. 2017). Other models addressing simultaneous design and dispatch with combined heat and power tend to produce sub-optimal solutions to the monolith (Blackburn et al. 2019), or optimal solutions to a problem with reduced scope (Burer et al. 2003; Buoro et al. 2014; Pruitt et al. 2013b; Silvente et al. 2015; Weber et al. 2006). In particular, although Pruitt et al. (2013a, b) make optimal design and dispatch decisions for a combined heat and power system, the pricing structure and operational details of the technologies are not as sophisticated as those we consider.

The Distributed Energy Resources-Customer Adoption Model (DER- CAM) is a mixed-integer linear program. Authors such as Siddiqui et al. (2005), Stadler et al. (2014), Braslavsky et al. (2015), and Mashayekh et al. (2017) report on its capabili-

ties, which consist of generating an optimal design and dispatch strategy for a suite of technologies, subject to constraints on load shifting, peak shaving, power export agreements, and ancillary service markets. Some versions of this model consider detailed electrical distribution, but loads are not incorporated at a level as fine as hourly, the technologies are dispatched in a coarser manner, and the model lacks certain economic nuances, such as tiered monthly demand charges, minimum required utility payments, and offtake agreements. DESOD (Bracco et al. 2016) optimizes an energy system with CHP for independent buildings, omitting connections to the utility for thermal energy. Dispatch is also on a coarser level, e.g., by considering "typical-day" loads. A shortened time horizon expedites solutions. BALMOREL (Wiese et al. 2018) is an open-source model that considers thermal-producing and distributed energy generation (Karlsson and Meibom 2008; Koivisto et al. 2019; Nasution et al. 2019). Although the model possesses hourly fidelity, it fails to include resiliency and investment incentives. Connolly et al. (2010) and Ringkjøb et al. (2018) provide significant reviews of energy and electricity system analysis.

## 3 Mathematical formulation

We introduce the monolith mixed-integer linear programming formulation of our design and dispatch problem,  $(\hat{\mathcal{R}})$ . This model is an extension of that given in Ogunmoded et al. (2021), which we term ( $\mathcal{R}$ ), and introduces combined heat and power into the system. Figure 2 summarizes the variables, objectives, and constraints of the monolith, which seeks design and dispatch decisions for a system of distributed energy resources that minimizes the cost of capital, operations and maintenance (O&M), fuel and utility costs, net of production incentives and energy exports. Constraints ensure that: (i) system sizing and fuel consumption fall within user-specified limits, (ii) pro-



**Fig. 2** A categorical overview of the decision variables, objective function, and constraints that compose the REopt Lite optimization model,  $(\hat{\mathcal{R}})$ , where SOC denotes state of charge

duction and load balance in each time period, and (iii) production incentives, utility charges, and other policy structures are accurately accounted for.

This section presents contributions to the model that involve additional or significantly altered constraints. We provide first notation used for these additions, in alphabetic order, and categorized as: (i) indices and sets, (ii) parameters, and (iii) variables. We state for ease of exposition the objective function, and then give the sets of constraints that were significantly modified from ( $\mathcal{R}$ ). Finally, we point to the appendix for the remainder, which we include for completeness. Our naming convention represents sets using calligraphic capital letters, parameters employing lower-case letters, and variables invoking upper-case letters. Subscripts denote indices, whereas superscripts and other "decorations" represent similar constructs with the same "stem."

#### 3.1 Sets and parameters

Sets	
B	Storage systems
$\mathcal{C}$	Technology classes
$\mathcal{D}$	Time-of-use demand periods
ε	Electrical time-of-use demand tiers
${\mathcal F}$	Fuel types
$\mathcal{H}$	Time steps
$\mathcal{M}$	Months of the year
$\mathcal{N}$	Monthly peak demand tiers
$\mathcal{T}$	Technologies
U	Total electrical energy pricing tiers
Subsets and indexed	sets
$\mathcal{B}^c \subseteq \mathcal{B}^{th}$	Cold thermal energy storage systems
$\mathcal{B}^{e} \subseteq \mathcal{B}$	Electrical storage systems
$\mathcal{B}^{\mathrm{h}} \subseteq \mathcal{B}^{\mathrm{th}}$	Hot thermal energy storage systems
$\mathcal{B}^{\mathrm{th}} \subseteq \mathcal{B}$	Thermal energy storage systems
$\mathcal{H}^g \subseteq \mathcal{H}$	Time steps in which grid purchasing is available
$\mathcal{H}_m \subseteq \mathcal{H}$	Time steps within a given month <i>m</i>
$\mathcal{H}_d \subseteq \mathcal{H}$	Time steps within electrical power time-of-use demand tier $d$
$\mathcal{K}_t \subseteq \mathcal{K}$	Subdivisions applied to technology t
$\mathcal{K}^{c} \subseteq \mathcal{K}$	Capital cost subdivisions
$S_{tk} \subseteq S$	Power rating segments from subdivision $k$ applied to technology $t$
$\mathcal{T}_b \subseteq \mathcal{T}$	Technologies that can charge storage system b
$\mathcal{T}_c \subseteq \mathcal{T}$	Technologies in class c
$\mathcal{T}_f \subseteq \mathcal{T}$	Technologies that burn fuel type $f$
$\mathcal{T}_v \subseteq \mathcal{T}$	Technologies that may access net-metering regime $v$
$\mathcal{T}^{\mathrm{ac}} \subseteq \mathcal{T}^{\mathrm{cl}}$	Absorption chillers
$\mathcal{T}^{CHP} \subseteq \mathcal{T}^{f}$	CHP technologies
$\mathcal{T}^{cl} \subseteq \mathcal{T}$	Cooling technologies
$\mathcal{T}^e \subseteq \mathcal{T}$	Electricity-producing technologies
$\mathcal{T}^{ec} \subseteq \mathcal{T}^{cl}$	Electric chillers
$\mathcal{T}^f \subseteq \mathcal{T}^e$	Fuel-burning, electricity-producing technologies
$\mathcal{T}^{ht} \subseteq \mathcal{T}$	Heating technologies
$\mathcal{T}^{\mathrm{td}} \subseteq \mathcal{T}$	Technologies that cannot turn down, i.e., PV and wind
$\mathcal{U}^{\mathbf{p}} \subset \mathcal{U}$	Electrical energy purchase pricing tiers
$\mathcal{U}^{\mathrm{s}}_{t} \subset \mathcal{U}^{\mathrm{s}}$	Electrical energy sales pricing tiers accessible by technology t
$\mathcal{U}^{sb} \subset \mathcal{U}^{s}$	Electrical energy sales pricing tiers accessible by storage
	Electrical energy sales prioring does accessible of storage

[kW]

Scaling pa	arameters	
Г	Number of time periods within a day	[-]
Δ	Time step scaling	[h]
$\Theta_{M}$	Peak load oversizing factor	[_]
M		[various]
Parameter	rs for costs and their functional forms	
c <sup>afc</sup>	Utility annual fixed charge	[\$]
cts cm	y-intercept of capital cost curve for technology $t$ in segment $s$	[\$]
$c_{ts}^{\text{cm}}$	Slope of capital cost curve for technology <i>t</i> in segment <i>s</i>	[\$/kW]
$c_{uh}^{e}$	Export rate for energy in energy demand tier $u$ in time step $h$	[\$/kWh]
$c_{uh}^{g}$	Grid energy cost in energy demand tier $u$ during time step $h$	[\$/kWh]
$c_b^{\rm kW}$	Capital cost of power capacity for storage system b	[\$/kW]
$c_b^{\rm kWh}$	Capital cost of energy capacity for storage system b	[\$/kWh]
$c_b^{\mathrm{omb}}$	Operation and maintenance cost of storage system <i>b</i> per unit of energy rating	[\$/kWh]
$c_t^{\text{omp}}$	Operation and maintenance cost of technology <i>t</i> per unit of production	[\$/kWh]
$c_t^{\text{om}\sigma}$	Operation and maintenance cost of technology <i>t</i> per unit of power rating, including standby charges	[\$/kW]
$c_{de}^{\rm r}$	Cost per unit peak demand in time-of-use demand period $d$ and tier $e$	[\$/kW]
$c_{mn}^{\rm rm}$	Cost per unit peak demand in tier $n$ during month $m$	[\$/kW]
$c_f^{u}$	Unit cost of fuel type $f$	[\$/MMBTU]
Demand p	parameters	
$\delta_h^c$	Cooling load in time step h	[kW]
$\delta_h^{\mathrm{d}}$	Electrical load in time step h	[kW]
$\bar{\delta}_{u}^{\mathrm{gs}}$	Maximum allowable sales in electrical energy demand tier $u$	[kWh]
$\delta_h^{\rm h}$	Heating load in time step h	[kW]
$\delta^{lp}$	Look-back proportion for ratchet charges	[fraction]
$\bar{\delta}_n^{\mathrm{mt}}$	Maximum monthly electrical power demand in peak pricing tier $n$	[kW]
$\bar{\delta}_{e}^{\mathrm{t}}$	Maximum power demand in time-of-use demand tier e	[kW]
$\bar{\delta}_{u}^{\mathrm{tu}}$	Maximum monthly electrical energy demand in tier $u$	[kWh]
Incentive	parameters	
ī	Upper incentive limit for technology <i>t</i>	[\$]
i <sub>v</sub> <sup>n</sup>	Net metering limits in net metering regime $v$	[kW]
$i_t^r$	Incentive rate for technology <i>t</i>	[\$/kWh]

Technology-specific time-series factor parameters	

technology t

$f_{th}^{\rm ed}$	Electrical power derate factor of technology $t$ at time step $h$	[unitless]
$f_{th}^{\mathrm{fa}}$	Fuel burn ambient correction factor of technology $t$ at time step $h$	[unitless]
$f_{th}^{\text{ha}}$	Hot water ambient correction factor of technology $t$ at time step $h$	[unitless]

Maximum power rating for obtaining production incentive for

 $\bar{\imath}_t^{\sigma}$ 

$f_{th}^{\rm ht}$	Hot water thermal grade correction factor of technology $t$ at time step $h$	[unitless]
$f_{th}^{\rm p}$	Production factor of technology $t$ during time step $h$	[unitless]
Technology	-specific factor parameters	
$f_t^d$	Derate factor for turbine technology <i>t</i>	[unitless]
$f_t^{l}$	Levelization factor of technology t	[fraction]
$f_t^{\text{li}}$	Levelization factor of production incentive for technology t	[fraction]
$f_t^{\rm pf}$	Present worth factor for fuel for technology t	[unitless]
$f_t^{\rm pi}$	Present worth factor for incentives for technology t	[unitless]
$f_t^{\text{td}}$	Minimum turn down for technology <i>t</i>	[unitless]
Generic fac	tor parameters	
f <sup>e</sup>	Energy present worth factor	[unitless]
$f^{\text{om}}$	O&M present worth factor	[unitless]
$f^{\text{tot}}$	Tax rate factor for off-taker	[fraction]
$f^{\text{tow}}$	Tax rate factor for owner	[fraction]
Power ratin	g and fuel limit parameters	
$b_f^{\mathrm{fa}}$	Amount of available fuel for fuel type $f$	[MMBTU]
$\bar{b}_t^{\sigma}$	Maximum power rating for technology t	[kW]
Efficiency p	arameters	
$\eta_{bt}^+$	Efficiency of charging storage system $b$ using technology $t$	[fraction]
$\eta_b^-$	Efficiency of discharging storage system b	[fraction]
$\eta^{ac}$	Absorption chiller efficiency	[fraction]
$\eta^{D}$	Boiler efficiency	[fraction]
$\eta^{ee}_{g+}$	Electric chiller efficiency	[fraction]
ησ	Efficiency of charging electrical storage using grid power	[IIaction]
Storage par	cameters	
$\bar{w}_b^{ m bkW}$	Maximum power output of storage system b	[kW]
$\underline{w}_{b}^{bkW}$	Minimum power output of storage system b	[kW]
$\bar{w}_b^{\rm bkWh}$	Maximum energy capacity of storage system b	[kWh]
$\underline{w}_{b}^{bkWh}$	Minimum energy capacity of storage system b	[kWh]
$w_b^{\mathrm{d}}$	Decay rate of storage system b	[1/h]
$\underline{w}_{h}^{\text{mcp}}$	Minimum percent state of charge of storage system b	[fraction]
$w_b^0$	Initial percent state of charge of storage system b	[fraction]
Fuel burn p	arameters	
mfb	v-intercept of the fuel rate curve for technology t	[MMBTU/h]
m <sup>fbm</sup>	Fuel burn rate v-intercept per unit size for technology $t$	[MMBTU/kWh]
$m_t^{fm}$	Slope of the fuel rate curve for technology <i>t</i>	[MMBTU/kWh]
CHP therm	al performance parameters	
$k_t^{\text{te}}$	Thermal energy production of CHP technology $t$ per unit electrical	[unitless]
, tp	output	r (d. 1
K <sub>t</sub> '	rating	[unitless]

# 3.2 Variables

Boundary conditions

$X_{b,0}^{se}$	Initial state of charge for storage system b	[kWh]
Continuo	us variables	
$X_{h}^{bkW}$	Power rating for storage system b	[kW]
$X_{b}^{bkWh}$	Energy rating for storage system b	[kWh]
$X_{de}^{de}$	Peak electrical power demand allocated to tier $e$ and time-of-use demand period $d$	[kW]
$X_{bh}^{dfs}$	Power discharged from storage system $b$ during time step $h$	[kW]
$X_{mn}^{dn}$	Peak electrical power demand allocated to tier $n$ during month $m$	[kW]
$X_{th}^{f}$	Fuel burned by technology $t$ in time step $h$	[MMBTU/h]
$X_{th}^{\text{fb}}$	y-intercept of fuel burned by technology $t$ in time step $h$	[MMBTU/h]
$X_{uh}^{g}$	Power purchased from the grid for electrical load in demand tier $u$ during time step $h$	[kW]
$X_{h}^{\text{gts}}$	Electrical power delivered to storage by the grid in time step $h$	[kW]
$X^{mc}$	Annual utility minimum charge adder	[\$]
$X_t^{\rm pi}$	Production incentive collected for technology t	[\$]
X <sup>plb</sup>	Peak electrical demand during look back periods	[kW]
$X_{tuh}^{\rm ptg}$	Exports from production to the grid by technology $t$ in demand tier $u$ during time step $h$	[kW]
$X_{bth}^{\text{pts}}$	Power from technology $t$ used to charge storage system $b$ during time step $h$	[kW]
$X_{th}^{\text{ptw}}$	Thermal power from technology $t$ sent to waste or curtailed during time step $h$	[kW]
$X_{th}^{rp}$	Rated production of technology $t$ during time step $h$	[kW]
$X_t^{\sigma}$	Power rating of technology t	[kW]
$X_{tks}^{\sigma s}$	Power rating of technology $t$ allocated to subdivision $k$ , segment $s$	[kW]
$X_{bh}^{se}$	State of charge of storage system $b$ at the end of time step $h$	[kWh]
$X_{uh}^{stg}$	Exports from storage to the grid in demand tier $u$ during time step $h$	[kW]
$X_{th}^{tp}$	Thermal production of technology $t$ in time step $h$	[kW]
$X_{th}^{tpb}$	y-intercept of thermal production of CHP technology $t$ in time step $h$	[kW]
Binary vo	iriables	
$Z_{tks}^{\sigma s}$	1 If technology $\overline{t}$ in subdivision $k$ , segment $s$ is chosen; 0 otherwise	[unitless]
$Z_{th}^{to}$	1 If technology $t$ is operating in time step $h$ ; 0 otherwise	[unitless]

#### 3.3 Objective function

$$\begin{aligned} &(\widehat{\mathcal{R}}) \quad \mininimize \underbrace{\sum_{t \in \mathcal{T}, k \in \mathcal{K}^{c}, s \in \mathcal{S}_{tk}} \left( c_{ts}^{cm} \cdot X_{tks}^{\sigma s} + c_{ts}^{cb} \cdot Z_{tks}^{\sigma s} \right)}_{\text{Generating Technology Capital Costs}} \\ &+ \underbrace{\sum_{b \in \mathcal{B}} \left( c_{b}^{kW} \cdot X_{b}^{bkW} + \left( c_{b}^{kWh} + c_{b}^{omb} \right) \cdot X_{b}^{bkWh} \right)}_{\text{Storage Capital Costs}} \\ &+ (1 - f^{tow}) \cdot f^{om} \cdot \left( \sum_{t \in \mathcal{T}} c_{t}^{om\sigma} \cdot X_{t}^{\sigma} + \sum_{t \in \mathcal{T}^{t}, h \in \mathcal{H}} c_{t}^{omp} \cdot X_{th}^{rp} \right) \\ &+ (1 - f^{tot}) \cdot \Delta \cdot \sum_{f \in \mathcal{F}} c_{t}^{p} \cdot \sum_{t \in \mathcal{T}_{f, h} \in \mathcal{H}} f_{t}^{pf} \cdot X_{th}^{g} \\ &+ (1 - f^{tot}) \cdot f^{e} \cdot \left( \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} c_{uh}^{g} \cdot X_{uh}^{g} \right) \\ &+ (1 - f^{tot}) \cdot f^{e} \cdot \left( \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} c_{uh}^{g} \cdot X_{uh}^{g} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot f^{e} \cdot \left( \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} c_{uh}^{g} \cdot X_{uh}^{g} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot f^{e} \cdot \left( \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} c_{uh}^{g} \cdot X_{uh}^{g} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot f^{e} \cdot \left( \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} c_{uh}^{g} \cdot X_{uh}^{g} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot f^{e} \cdot \left( \sum_{u \in \mathcal{U}^{b}, h \in \mathcal{H}^{g}} c_{uh}^{e} + \sum_{m \in \mathcal{M}, n \in \mathcal{N}} c_{mn}^{rm} \cdot X_{mn}^{dn} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot \sum_{h \in \mathcal{H}^{g}} c_{uh}^{e} \cdot X_{uh}^{stg} + \sum_{t \in \mathcal{T}, u \in \mathcal{U}_{t}^{g}} c_{uh}^{e} \cdot X_{tuh}^{th} \right) \\ &+ \left( 1 - f^{tot} \right) \cdot \sum_{h \in \mathcal{H}^{g}} \sum_{u \in \mathcal{U}^{a}} c_{uh}^{e} \cdot X_{uh}^{stg} + \sum_{t \in \mathcal{T}, u \in \mathcal{U}_{t}^{g}} c_{uh}^{e} \cdot X_{tuh}^{th} \right) \\ &+ \left( 1 - f^{tow} \right) \cdot \sum_{h \in \mathcal{H}^{g}} \sum_{u \in \mathcal{U}^{a}} \sum_$$

The objective function is the same as that in  $(\mathcal{R})$  and minimizes energy life cycle cost, i.e., capital costs, O&M costs, and utility costs; it maximizes (by subtracting) payments for energy exports and other incentives.

# **3.4 Constraints**

We mathematically present and describe the constraints that we modify from Ogunmodede et al. (2021) to account for combined heat and power. For ease of presentation, we exclude initial conditions and other minor exceptions.

#### 3.4.1 Fuel constraints

$$\Delta \cdot \sum_{t \in \mathcal{I}_f, h \in \mathcal{H}} X_{th}^{\mathrm{f}} \le b_f^{\mathrm{fa}} \quad \forall f \in \mathcal{F}$$
(1a)

$$X_{th}^{\rm f} = m_t^{\rm fm} \cdot f_{th}^{\rm p} \cdot X_{th}^{\rm rp} + m_t^{\rm fb} \cdot Z_{th}^{\rm to} \quad \forall t \in \mathcal{T}^{\rm f} \backslash \mathcal{T}^{\rm CHP}, h \in \mathcal{H}$$
(1b)

$$X_{th}^{f} = m_{t}^{fm} \cdot X_{th}^{tp} \quad \forall t \in \mathcal{T}^{ht} \setminus \mathcal{T}^{CHP}, \ h \in \mathcal{H}$$
(1c)

$$X_{th}^{f} = f_{th}^{fa} \cdot \left( X_{th}^{fb} + f_{th}^{p} \cdot m_{t}^{fm} \cdot X_{th}^{rp} \right) \quad \forall t \in \mathcal{T}^{CHP}, h \in \mathcal{H}$$
(1d)

$$m_t^{\text{fbm}} \cdot X_t^{\sigma} - M \cdot (1 - Z_{th}^{\text{to}}) \le X_{th}^{\text{fb}} \quad \forall t \in \mathcal{T}^{\text{CHP}}, h \in \mathcal{H}$$
(1e)

Constraints (1) enforce the fuel requirements for the combustion-powered technologies in REopt Lite. Constraint (1a) limits the available quantity of each fuel type per annum; we assume that while multiple technologies (e.g., natural gas boilers and CHP systems) may share the same fuel type, each technology may burn at most one type of fuel. Constraint (1b) enforces both a fixed consumption rate per hour of operational time and a variable burn rate per unit of energy produced for each electric, non-CHP technology. Constraint (1c) applies logic similar to that in constraint (1b) for non-CHP heating technologies, but removes the fixed fuel consumption rate during operation. Constraint (1d) defines fuel consumption for CHP systems using both a per-operating-hour rate and a per-unit-production rate, but, unlike constraint (1b), the hourly burn rate is a decision variable; constraint (1e) sets this decision variable to a fixed proportion of the system's power rating if it is operating, and to zero otherwise.

#### 3.4.2 Thermal production constraints

$$X_{th}^{\text{tpb}} \le \min\left\{k_t^{\text{tp}} \cdot X_t^{\sigma}, M \cdot Z_{th}^{\text{to}}\right\} \quad \forall t \in \mathcal{T}^{\text{CHP}}, h \in \mathcal{H}$$
(2a)

$$X_{th}^{\text{tpb}} \ge k_t^{\text{tp}} \cdot X_t^{\sigma} - M \cdot (1 - Z_{th}^{\text{to}}) \quad \forall t \in \mathcal{T}^{\text{CHP}}, h \in \mathcal{H}$$
(2b)

$$f_{th}^{\text{ha}} \cdot f_{th}^{\text{ht}} \cdot \left(k_t^{\text{te}} \cdot f_{th}^{\text{p}} \cdot X_{th}^{\text{rp}} + X_{th}^{\text{tpb}}\right) = X_{th}^{\text{tp}} \quad \forall t \in \mathcal{T}^{\text{CHP}}, h \in \mathcal{H}$$
(2c)

Constraints (2a)–(2b) limit the fixed component of thermal production of CHP technology t in time step h to the product of the thermal power production per unit of power rating and the power rating itself if the technology is operating, and 0 if it is not. Constraint (2c) relates the thermal production of a CHP technology to its constituent components, where the relationship includes a term that is proportional to electrical power production in each time step.

#### 3.4.3 Storage system constraints

Boundary Conditions and Size Limits

$$X_{b,0}^{\text{se}} = w_b^0 \cdot X_b^{\text{bkWh}} \quad \forall b \in \mathcal{B}$$
(3a)

$$\underline{w}_{b}^{\text{bkWh}} \leq X_{b}^{\text{bkWh}} \leq \bar{w}_{b}^{\text{bkWh}} \quad \forall b \in \mathcal{B}$$
(3b)

$$\underline{w}_{b}^{\text{bkW}} \le X_{b}^{\text{bkW}} \le \bar{w}_{b}^{\text{bkW}} \quad \forall b \in \mathcal{B}$$
(3c)

Constraint (3a) initializes a storage system's state of charge using a fraction of its energy rating; constraints (3b) and (3c) limit the storage system size under the implicit assumption that a storage system's power and energy ratings are independent. These constraints are identical to those given in ( $\mathcal{R}$ ), but work in conjunction with significantly modified storage constraints that directly follow.

Storage Operations

$$X_{bth}^{\text{pts}} + \sum_{u \in \mathcal{U}_t^{\text{s}}} X_{tuh}^{\text{ptg}} \le f_{th}^{\text{p}} \cdot f_t^{\text{l}} \cdot X_{th}^{\text{rp}} \quad \forall b \in \mathcal{B}^{\text{e}}, \ t \in \mathcal{T}^{\text{e}}, \ h \in \mathcal{H}^{\text{g}}$$
(3d)

$$X_{bth}^{\text{pts}} \le f_{th}^{\text{p}} \cdot f_{t}^{\text{l}} \cdot X_{th}^{\text{rp}} \quad \forall b \in \mathcal{B}^{\text{e}}, \ t \in \mathcal{T}^{\text{e}}, \ h \in \mathcal{H} \setminus \mathcal{H}^{\text{g}}$$
(3e)

$$X_{bth}^{\text{pts}} \le f_{th}^{\text{p}} \cdot X_{th}^{\text{tp}} \quad \forall b \in \mathcal{B}^{\text{th}}, \ t \in \mathcal{T}_b \setminus \mathcal{T}^{\text{CHP}}, \ h \in \mathcal{H}$$
(3f)

$$X_{bth}^{\text{pts}} + X_{th}^{\text{ptw}} \le X_{th}^{\text{tp}} \quad \forall b \in \mathcal{B}^{\text{h}}, \ t \in \mathcal{T}^{\text{CHP}}, \ h \in \mathcal{H}$$
(3g)

$$X_{bh}^{se} = X_{b,h-1}^{se} + \Delta \cdot \left( \sum_{t \in \mathcal{T}^e} (\eta_{bt}^+ \cdot X_{bth}^{pts}) + \eta^{g_+} \cdot X_h^{gts} - X_{bh}^{dfs} / \eta_b^- \right)$$
$$\forall b \in \mathcal{B}^e, \ h \in \mathcal{H}^g \tag{3h}$$

$$X_{bh}^{se} = X_{b,h-1}^{se} + \Delta \cdot \left( \sum_{t \in \mathcal{T}^e} (\eta_{bt}^+ \cdot X_{bth}^{pts}) - X_{bh}^{dfs} / \eta_b^- \right)$$
$$\forall b \in \mathcal{B}^e, \ h \in \mathcal{H} \setminus \mathcal{H}^g$$
(3i)

$$X_{bh}^{se} = X_{b,h-1}^{se} + \Delta \cdot \left( \sum_{t \in \mathcal{T}_b} \eta_{bt}^+ \cdot X_{bth}^{pts} - X_{bh}^{dfs} / \eta_{\bar{b}}^- - w_b^d \cdot X_{bh}^{se} \right)$$
$$\forall b \in \mathcal{B}^{th}, h \in \mathcal{H}$$
(3j)

$$X_{bh}^{\text{se}} \ge \underline{w}_{b}^{\text{mcp}} \cdot X_{b}^{\text{bkWh}} \quad \forall b \in \mathcal{B}, h \in \mathcal{H}$$
(3k)

Constraints (3d) and (3e) restrict the electrical power that charges storage and is exported to the grid (in the former case), or that charges storage only (in the latter case, when grid export is unavailable) from each technology in each time step relative to the amount of electricity produced. Constraint (3f) provides an analogous restriction to that of constraint (3e) for thermal production, and constraint (3g) provides the same restriction for the thermal production of CHP systems. Constraints (3h), (3i), and (3j) balance state of charge for each storage system and time period for three specific cases, respectively: (i) available grid-purchased electricity, (ii) lack of grid-purchased

electricity, and (iii) thermal storage, in which we account for decay. Constraint (3k) ensures that minimum state-of-charge requirements are not violated.

**Charging Rates** 

$$X_{b}^{\text{bkW}} \ge \sum_{t \in \mathcal{T}_{b}} X_{bth}^{\text{pts}} + X_{h}^{\text{gts}} + X_{bh}^{\text{dfs}} \quad \forall b \in \mathcal{B}^{\text{e}}, h \in \mathcal{H}^{\text{g}}$$
(31)

$$X_{b}^{\text{bkW}} \ge \sum_{t \in \mathcal{T}_{b}} X_{bth}^{\text{pts}} + X_{bh}^{\text{dfs}} \quad \forall b \in \mathcal{B}^{\text{e}}, h \in \mathcal{H} \setminus \mathcal{H}^{\text{g}}$$
(3m)

$$X_{b}^{bkW} \ge \sum_{t \in \mathcal{T}_{b}} X_{bth}^{pts} + X_{bh}^{dfs} \quad \forall b \in \mathcal{B}^{th}, h \in \mathcal{H}$$
(3n)

$$X_{bh}^{\text{se}} \le X_{b}^{\text{bkWh}} \quad \forall b \in \mathcal{B}, h \in \mathcal{H}$$
(30)

Constraints (31) and (3m) require that a battery's power rating must meet or exceed its rate of charge or discharge; the latter constraint considers the case in which the grid is not available. Constraint (3n) reflects the power requirements for the thermal system. Constraint (3o) requires a storage system's energy level to be at or below the corresponding rating.

Cold and hot thermal loads

$$\sum_{t \in \mathcal{T}^{cl}} f_{th}^{p} \cdot X_{th}^{tp} + \sum_{b \in \mathcal{B}^{c}} X_{bh}^{dfs} = \delta_{h}^{c} \cdot \eta^{ec} + \sum_{b \in \mathcal{B}^{c}, t \in \mathcal{T}^{cl}} X_{bth}^{pts} \quad \forall h \in \mathcal{H}$$
(4a)  
$$\sum_{t \in \mathcal{T}^{CHP}} X_{th}^{tp} + \sum_{t \in \mathcal{T}^{ht} \setminus \mathcal{T}^{CHP}} f_{th}^{p} \cdot X_{th}^{tp} + \sum_{b \in \mathcal{B}^{h}} X_{bh}^{dfs} = \delta_{h}^{h} \cdot \eta^{b}$$
$$+ \sum_{t \in \mathcal{T}^{CHP}} X_{th}^{ptw} + \sum_{b \in \mathcal{B}^{h}, t \in \mathcal{T}^{ht}} X_{bth}^{pts} + \sum_{t \in \mathcal{T}^{ac}} X_{th}^{tp} / \eta^{ac} \quad \forall h \in \mathcal{H}$$
(4b)

Constraints (4a) and (4b) balance cold and hot thermal loads, respectively, by equating the power production and the power from storage with the sum of the demand, the power to storage, and, in the case of cold loads, from the absorption chillers as well. Here, for legacy reasons, we have scaled the power by the efficiency of the respective technology; based on our variable definitions, we could have equivalently adjusted these by a coefficient of performance.

#### 3.4.4 Production constraints

$$X_{th}^{\rm rp} \le \bar{b}_t^{\sigma} \cdot Z_{th}^{\rm to} \quad \forall t \in \mathcal{T}, h \in \mathcal{H}$$
(5a)

$$\underbrace{f}_{t}^{\mathrm{td}} \cdot X_{t}^{\sigma} - X_{th}^{\mathrm{rp}} \le \bar{b}_{t}^{\sigma} \cdot (1 - Z_{th}^{\mathrm{to}}) \quad \forall t \in \mathcal{T}, h \in \mathcal{H}$$
(5b)

$$X_{th}^{\text{tp}} \le X_t^{\sigma} \quad \forall t \in \mathcal{T} \setminus \mathcal{T}^{\text{e}}, h \in \mathcal{H}$$
(5c)

Deringer

Constraint set (5) ensures that the rated production lies between a minimum turndown threshold and a maximum system size; constraints (5a) and (5b) are copied from Ogunmodede et al. (2021), while constraint (5c) is new. Constraint (5a) restricts system power output to its rated capacity when the technology is operating, and to 0 otherwise. Constraint (5b) ensures a minimum power output while a technology is operating; otherwise, the constraint is dominated by simple bounds on production. Constraint (5c) ensures that the thermal production of non-CHP heating and cooling technologies does not exceed system size.

The remainder of the formulation largely mimics that of  $(\mathcal{R})$  given in Ogunmodede et al. (2021), and is provided in the appendix (along with additional notation).

### 4 Solution methodology

The mathematical formulation in Section 3 extends the model given in Ogunmodede et al. (2021) to incorporate combined heat and power, which entails the introduction of more technologies and the corresponding constraints to control them, including balancing multiple loads, i.e., cold thermal, hot thermal, and electrical. As such, instances of  $(\hat{\mathcal{R}})$  are more difficult to solve. In order to improve tractability and to enable the model's use in the web-based tool described in Mishra et al. (2021), we reformulate the model by: (i) introducing tailored data structures; (ii) efficiently handling data; and, (iii) reformulating with a more streamlined set of variables. The improvements are made relative to the implementation of the formulation given in Cutler et al. (2017), and which we term  $(\hat{\mathcal{R}})$ .

#### 4.1 Introducing tailored data structures

Reducing the instantiation of parameters and variables through the judicious use of sets is critical to mitigating otherwise large instances who size would preclude them from being solved in a practical amount of time. Brown and Dell (2007) (§4) point towards small examples, while Klotz and Newman (2013b) explain theoretical and computational difficulties associated with large models. Formulation ( $\hat{\mathcal{R}}$ ) employs subsets and indexed sets to ensure that only appropriate decision variables appear in the objective function and in the constraints, either as a sum and/or according to a constraint qualifier. This reduces the size of the model, both in terms of the number of variables and in terms of the number of constraints an instance contains.

For example, the set  $\mathcal{T}$  represents all available technologies; the subset  $\mathcal{T}^{e}$  contains only electricity producing technologies. Similarly,  $\mathcal{B}$  represents the set of storage systems; the subset  $\mathcal{B}^{e}$  contains only electrical energy storage systems. Constraint (3d) highlights the efficiency of using subsets to limit electrical power dispatched to charge storage and export to the grid for electricity-producing technologies and storage systems.

$$X_{bth}^{\text{pts}} + \sum_{u \in \mathcal{U}_t^{\text{s}}} X_{tuh}^{\text{ptg}} \le f_{th}^{\text{p}} \cdot f_t^{\text{l}} \cdot X_{th}^{\text{rp}} \quad \forall b \in \mathcal{B}^{\text{e}}, \ t \in \mathcal{T}^{\text{e}}, \ h \in \mathcal{H}^{\text{g}}$$
(3d)

The original formulation used sets, rather than subsets, controlling the terms that appeared in the constraints using binary indicator parameters. Not only did this introduce unnecessary parameters, it induced more terms in the constraint, and an increased number of constraints overall.

A construct similar to subsets that also limits the number of variables and constraints appearing in a model instance is *indexed sets* that restrict the size of a set to relevant elements based on another set. Constraint (3f), reformulated from the original model, provides an example in which the set of technologies considered is restricted to those associated with storage, resulting in a qualifier  $T_b$ , rather than a constraint based on each element in the entire set of technologies, T:

$$X_{bth}^{\text{pts}} \le f_{th}^{\text{p}} \cdot X_{th}^{\text{tp}} \quad \forall b \in \mathcal{B}^{\text{th}}, \ t \in \mathcal{T}_b \setminus \mathcal{T}^{\text{CHP}}, \ h \in \mathcal{H}$$
(3f)

Similar to the case in which subsets replace larger sets, the case in which indexed sets replace larger sets precludes the need for binary parameters, and reduces both the number of terms a constraint contains, and the number of constraints overall in a given formulation instance.

#### 4.2 Efficiently handling data

Data used to populate  $(\widehat{\mathcal{R}})$  are drawn from myriad sources, including user-specified inputs, and had been introduced into the original model at disjoint stages during its development, resulting in (i) complex calculations to determine various parameter values; (ii) superfluous variables representing calculations consisting of both parameters and variables; and, (iii) arbitrarily high variable bounds, resulting in potential numerical stability issues (in the case of simple bounds, see Klotz and Newman (2013a)), and in weak linear programming relaxations (in the case of big-M values, see Camm et al. (1990)). In order to preclude variables with unnecessarily and arbitrarily high bounds, we reduce the values using physical limitations, appropriate for a given model instance.

Constraint (1a) provides an example of a simple variable bound on  $X_{th}^{f}$ :

$$\Delta \cdot \sum_{t \in \mathcal{T}_f, h \in \mathcal{H}} X_{th}^{\mathrm{f}} \le b_f^{\mathrm{fa}} \quad \forall f \in \mathcal{F}$$
(1a)

The maximum fuel limit,  $b_f^{\text{fa}}$ , had been an arbitrarily large value by default in the original formulation. In order to reduce the size of the feasible region and to control the discrepancy in the orders of magnitude between the largest and smallest non-zero values in a problem instance (thus, reducing the potential for numerical stability issues), we suggest a reasonable default value based on physical attributes of the system:

$$b_{f}^{\text{fa}} = \sum_{h \in \mathcal{H}} f_{th}^{\text{fa}} \cdot m_{t}^{\text{fbm}} \cdot \bar{b}_{t}^{\sigma} + f_{th}^{\text{fa}} \cdot f_{th}^{P} \cdot m_{t}^{\text{fm}} \cdot \bar{b}_{t}^{\sigma} \quad t \in \mathcal{T}^{\text{CHP}}$$

$$b_{f}^{\text{fa}} = |\mathcal{H}| \cdot m_{t}^{\text{fm}} \cdot \bar{b}_{t}^{\sigma} \quad \forall t \in \mathcal{T}^{\text{ht}} \setminus \mathcal{T}^{\text{CHP}}$$

$$b_{f}^{\text{fa}} = \sum_{h \in \mathcal{H}} m_{t}^{\text{fm}} \cdot f_{th}^{P} \cdot \bar{b}_{t}^{\sigma} + m_{t}^{\text{fb}} \quad \forall t \in \mathcal{T}^{f} \setminus (\mathcal{T}^{\text{ht}} \cup \mathcal{T}^{\text{CHP}})$$

🖄 Springer

where the first constraint pertains to a CHP technology, the second to a boiler, and the third to a diesel generator, respectively.

An example that illustrates a reduction in big-M values follows. Constraint (7a) permits nonzero power ratings only for the selected technology and corresponding subdivision in each class:

$$X_t^{\sigma} \le M \cdot \sum_{s \in \mathcal{S}_{tks}} Z_{tks}^{\sigma s} \quad \forall c \in \mathcal{C}, t \in \mathcal{T}_c, k \in \mathcal{K}_t$$
(7a)

Here, the big-M value for the system size can be given by the product of the number of hours in a day and the peak hourly load, assuming there is no economic incentive for exporting energy greater than the peak hourly load; this quantity would conservatively meet daily load, i.e., in the absence of other technologies, including storage devices:

$$M = \bar{b}_t^{\sigma} = \max_{h \in \mathcal{H}} \{ 24 \cdot \delta_h^{\mathsf{d}} \}$$

Table 1 highlights the list of values we tailor throughout formulation  $(\widehat{\mathcal{R}})$ .

#### 4.3 Reformulation with streamlined variables

Mixed-integer programs can assume a variety of mathematically equivalent formulations. However, some render instances that are more easily solved than others, in large part owing not to obvious theoretical characteristics, but to a practitioner's understanding of a solver's ability to exploit certain mathematical structures (Trick 2005). It is in this spirit that we examine  $(\bar{\mathcal{R}})$  for possible improvements in the mathematical formulation. Specifically, in the original formulation,  $(\bar{\mathcal{R}})$ , a set  $j \in \mathcal{J}$  informed destinations for electrical power, e.g., site demand, storage, curtailment, or the grid, and determined the relevance of a technology for a particular constraint. Correspondingly, a decision variable,  $\hat{Y}_{ijhsu}^{rp}$ , appearing in the original model represented the rated production of technology *t* at destination *j* during time step *h* in segment *s* from pricing tier *u*, in which elements *t* in the set  $\mathcal{T}$  represented all dispatchable technologies, including electricity from the grid. Separately, the decision variable  $\hat{Y}_{jheun}^{g}$  was defined as electrical power from the grid dispatched to destination *j*, in time step *h*, for demand bin *e*, in pricing tier *u*, and monthly peak demand tier *n*, and a constraint ensured that rated production by the utility was equal to grid purchases:

$$\sum_{s \in \mathcal{S}} \hat{Y}_{tjhsu}^{rp} = \sum_{e \in \mathcal{E}, n \in \mathcal{N}} \hat{Y}_{jheun}^{g} \quad \forall j \in \mathcal{J}, u \in \mathcal{U}, h \in \mathcal{H}, t \in \mathcal{T} : t = `Grid'$$

Formulation  $(\widehat{\mathcal{R}})$  assumes that electricity from the grid has unlimited availability, removing the need for the above-mentioned constraint. In turn, we note that we can eliminate sets  $\mathcal{U}$ ,  $\mathcal{S}$ , and  $\mathcal{J}$  from a variable representing production for the following reasons, respectively: (i) grid purchases are represented by a separate decision variable, so we can excise the utility from the collection of technologies  $\mathcal{T}$ , and we can remove the utility-specific pricing tier index u from rated production; (ii) the set of segments the dotted line represent explicit right-hand-side "b"-values, while those below represent coefficients on binary variables that are either traditional big-M values or are potential replacements for said values based

Constraints	Input parameter	Tailored value	Bound rationale
(1a)	$b_f^{\mathrm{fa}}$	$\Delta \cdot \sum_{h \in \mathcal{H}} (f_{th}^{\text{fa}} \cdot m_t^{\text{fbm}} \cdot \bar{b}_t^{\sigma} + f_{th}^{\text{fa}} \cdot f_{th}^{\text{p}} \cdot m_t^{\text{fm}} \cdot \bar{b}_t^{\sigma})$	Per (1d) and (1e) <sup><math>\dagger</math></sup>
		$\Delta \cdot  \mathcal{H}  \cdot m_t^{\mathrm{fm}} \cdot ar{b}_t^{\sigma}$	Per $(1c)^{\dagger}$
		$\Delta \cdot (\sum_{h \in \mathcal{H}} m_t^{\text{fm}} \cdot f_{th}^{\text{p}} \cdot \bar{b}_t^{\sigma} +  \mathcal{H}  \cdot m_t^{\text{fb}})$	Per $(1b)^{\dagger}$
(3b)	$\bar{w}_{b}^{\mathrm{bkW}}$	$\frac{\bar{w}_b^{\text{bkWh}}}{\Delta}$	Practical
(3c)	$\bar{w}_b^{ m bkWh}$	$\sum_{h \in \mathcal{H}} \delta_h^d  \forall b \in \mathcal{B}^e$	Practical
		$\sum_{h \in \mathcal{H}} \delta_h^h  \forall b \in \mathcal{B}^h$	Practical
		$\sum_{h \in \mathcal{H}} \delta_h^c  \forall b \in \mathcal{B}^c$	Practical
(6b)	$\overline{\iota}_t^{\sigma}$	$\bar{b}_t^{\sigma}$	Practical
(8e), (8f)	$\bar{\delta}_u^{\mathrm{gs}}$	$\Delta \cdot \sum_{h \in \mathcal{H}} \delta_h^{\mathrm{d}}$	Practical
(1e)	М	$m_t^{ m fbm} \cdot ar{b}_t^\sigma$	Practical
(2a), (2b)	М	$ar{b}_t^{\sigma} \cdot k_t^{ ext{tp}}$	Practical
(5a), (5b), (7a)	$\bar{b}_t^{\sigma}$	$\max_{h \in \mathcal{H}} \{ \Gamma \cdot \delta_h^{\mathrm{d}} \}  \forall t \in \mathcal{T}^{\mathrm{e}}$	Practical
		$\max_{h \in \mathcal{H}} \{ \Theta \cdot \delta_h^h \}  \forall t \in \mathcal{T}^{\text{ht}} \setminus \mathcal{T}^{\text{CHP}}$	Practical
		$\max_{h \in \mathcal{H}} \{ \Theta \cdot \delta_h^c \}  \forall t \in \mathcal{T}^{ec}$	Practical
		$\sum_{h \in \mathcal{H}} \delta_h^{c} + \frac{\bar{w}_b^{b \text{kWh}}}{\Delta}  \forall t \in \mathcal{T}^{\text{ac}}$	Practical
(6a)	$\overline{\iota}_t$	$\sum_{h \in \mathcal{H}} \Delta \cdot i_t^{\mathbf{r}} \cdot f_t^{\mathbf{p}\mathbf{i}} \cdot f_{th}^{\mathbf{p}} \cdot f_t^{\mathbf{l}} \cdot \bar{b}_t^{\sigma}$	Practical
( <b>6b</b> )	М	$ar{b}^{\sigma}_t - ar{\iota}^{\sigma}_t$	Practical
(9b)	i <sup>n</sup> <sub>v</sub>	$\sum_{t \in \mathcal{T}_{\mathcal{V}}} f_t^{\mathbf{d}} \cdot \bar{b}_t^{\sigma}$	Per net metering limit
(10a)	$ar{\delta}^{ ext{tu}}_{ \mathcal{U} }$	$\max_{m \in \mathcal{M}} \left\{ \sum_{h \in \mathcal{H}_m} \delta_h^d \right\}$	Practical
(11a)	$ar{\delta}^{mt}_{ \mathcal{N} }$	$\max_{m \in \mathcal{M}} \left\{ \sum_{h \in \mathcal{H}_m} \delta_h^d \right\}$	Practical
(12a)	$\bar{\delta}^t_{ \mathcal{E} }$	$\max_{d \in \mathcal{D}} \left\{ \sum_{h \in \mathcal{H}_d} \delta_h^d \right\}$	Practical

<sup>†</sup>The tailored values hold for all relevant instances as given by the constraint qualifiers, e.g., the first expression in the third column of the table is valid  $\forall t \in \mathcal{T}^{CHP}$ , the second  $\forall t \in \mathcal{T}^{h} \setminus \mathcal{T}^{CHP}$  and the third  $\forall t \in \mathcal{T}^{f} \setminus (\mathcal{T}^{ht} \cup \mathcal{T}^{CHP})$ . Constraint qualifiers for the other expressions are either explicitly stated or are self-explanatory; some tailored values may be invariant by one or more indices, e.g., the tailored value for (8e) and (8f) holds  $\forall u \in \mathcal{U}$ 

on improvements to user-specified inputs

 $s \in S$  is only utilized for system sizing decisions; and, (iii) the set of destinations is now informed solely by the technology type. This results in the transformation of the more complicated variable  $\hat{Y}_{tjhsu}^{rp}$  into the much simplified rated production variable,  $X_{th}^{rp}$ , where the latter variable is related to system size as follows:

$$X_t^{\sigma} \le \bar{b}_t^{\sigma} \cdot \sum_{s \in \mathcal{S}_{tk}} Z_{tks}^{\sigma s} \quad \forall c \in \mathcal{C}, t \in \mathcal{T}_c, k \in \mathcal{K}_t$$
(7a)

$$X_{th}^{\rm rp} = X_t^{\sigma} \quad \forall t \in \mathcal{T}^{\rm td}, h \in \mathcal{H}$$
(7b)

$$X_{th}^{\rm rp} \le f_{th}^{\rm ed} \cdot X_t^{\sigma} \quad \forall t \in \mathcal{T} \setminus \mathcal{T}^{\rm td}, h \in \mathcal{H}$$
(7c)

We replace the set of destinations  $\mathcal{J}$  with variables that represent power flows to the grid  $(X_{tuh}^{\text{ptg}})$  and to storage  $(X_{bth}^{\text{pts}})$ . The relationships between these variables are enforced as follows:

$$X_{bth}^{\text{pts}} + \sum_{u \in \mathcal{U}_t^{\text{s}}} X_{tuh}^{\text{ptg}} \le f_{th}^{\text{p}} \cdot f_t^{\text{l}} \cdot X_{th}^{\text{rp}} \quad \forall b \in \mathcal{B}^{\text{e}}, \ t \in \mathcal{T}^{\text{e}}, \ h \in \mathcal{H}^{\text{g}}$$
(3d)

$$X_{bth}^{\text{pts}} \le f_{th}^{\text{p}} \cdot f_{t}^{\text{l}} \cdot X_{th}^{\text{rp}} \quad \forall b \in \mathcal{B}^{\text{e}}, \ t \in \mathcal{T}^{\text{e}}, \ h \in \mathcal{H} \setminus \mathcal{H}^{\text{g}}$$
(3e)

And to remove destination *j* from the original variable  $\hat{Y}_{jheun}^{g}$  containing it, we employ a decision variable for the flow of electricity from the grid to storage  $(X_{h}^{gts})$ :

$$\sum_{u \in \mathcal{U}^{p}} X_{uh}^{g} \ge X_{h}^{gts} \quad \forall h \in \mathcal{H}^{g}$$
(8c)

Without the indices e and n, we enforce peak demand by billing period via constraints (11d) and (12d):

$$\sum_{n \in \mathcal{N}} X_{mn}^{\mathrm{dn}} \ge \sum_{u \in \mathcal{U}^{\mathrm{p}}} X_{uh}^{\mathrm{g}} \quad \forall m \in \mathcal{M}, h \in \mathcal{H}_{m}$$
(11d)

$$\sum_{e \in \mathcal{E}} X_{de}^{de} \ge \max_{h \in \mathcal{H}_d} \left\{ \sum_{u \in \mathcal{U}^{p}} X_{uh}^{g}, \delta^{lp} \cdot X^{plb} \right\} \quad \forall d \in \mathcal{D}$$
(12d)

Finally, constraints (8a) and (8b) balance load using the rated production and grid purchasing variables described above. Figure 3 provides a network flow representation of electrical load balancing under the reformulation, with PV and storage as the technologies, and a grid connection.

The revised formulation contains significantly fewer variables. In Fig. 3, the blue nodes represent sources and the red nodes represent destinations. While inflows and



Fig. 3 A network flow representation of electrical load balancing in the reformulated model, using a PV system and a battery as the on-site technologies

outflows are balanced for the technologies and utility, constraints (8a) and (8b) enforce flow balance for the node labeled "Site Load" in periods with utility connectivity and with an outage, respectively. Constraints (3d) and (3e) enforce nonnegative flows from PV to site load with and without a grid connection, respectively. Constraints (8c) and (8d) provide an analogous restriction on flows to site load from the utility and the battery, respectively.

One demonstration of the increased efficiency of our reformulation is the presence of at most one arc between each pair of nodes in Fig. 3, whereas in a representation of the previous formulation, each arc departing from PV would have had one copy for each segment *s*, and each arc departing from the utility would have had at least one additional copy due to the presence of both rated-production and grid-purchase decision variables for the utility, regardless of the number of redundant copies due to additional sets in both cases.

## 5 Data and results

We derive data for the 12 cases against which we evaluate our performance-enhancing techniques from a test set developed by the National Renewable Energy Laboratory. Each case contains combined heat and power and a boiler, and also some combination of solar photovoltaics, electric chillers, absorption chillers, along with different forms of energy storage such as batteries and thermal (hot and cold) energy storage; we solve for a year's worth of dispatch decisions at hourly fidelity. In order to fully exercise the model attributes formulated mathematically and explained in Section 3, the case studies differ in the building type, total electrical energy pricing tiers (|U|), monthly peak demand tiers (|N|), time-of-use demand periods (|D|), and whether standby charges apply. (We note that standby charges occur only if CHP technologies are not allowed to reduce peak demand; see constraint (11d) and constraint (12d).)

	$\frac{\text{Model}(\bar{\mathcal{R}})}{\text{Number of}}$		Model (Î	रे)			
			Reduction (%) in		Density of A-matrix $\cdot 10^3$ (%)		$\log_{10}(k/k')$
Case	variables	constraints	variables	constraints	Model $(\bar{\mathcal{R}})$	Model ( $\widehat{\mathcal{R}}$ )	$\operatorname{Model}(\bar{\mathcal{R}})-\operatorname{Model}(\widehat{\mathcal{R}})$
1	333,030	481,914	31.6	36.3	1.03	1.37	5
2	333,030	481,914	28.9	36.3	1.03	1.35	5
3	333,030	481,914	28.9	36.3	1.03	1.35	5
4	490,738	665,255	33.9	43.4	0.72	1.06	5
5	701,038	691,661	55.0	46.8	0.53	1.09	5
6	841,148	691,817	61.5	46.8	0.58	1.09	5
7	1,471,910	735,731	76.2	50.0	0.55	1.15	5
8	490,738	665,255	35.7	43.4	0.72	1.11	5
9	578,366	857,993	30.3	45.9	0.57	0.83	5
10	333,030	481,914	31.6	36.3	1.03	1.37	5
11	333,030	481,914	31.6	36.3	1.03	1.37	5
12	333,030	481,914	31.6	36.3	1.03	1.37	6

**Table 2** Problem statistics for the 12 cases on which we perform computational experiments comparing model  $(\hat{\mathcal{R}})$  and model  $(\hat{\mathcal{R}})$ 

<sup>1</sup>Represents the difference in orders of magnitude between the largest and smallest non-zero entries in the data, i.e., across the *A*-matrix, *b*-vector and *c*-vector, as calculated using  $\log_{10}(k/k')$ , where *k*=max {all entries in *A*-matrix, *b*-vector, *c*-vector} and *k'*=min {all entries in *A*-matrix, *b*-vector, *c*-vector}

We first present the results by comparing models  $(\widehat{\mathcal{R}})$  and  $(\overline{\mathcal{R}})$  in terms of their problem statistics. Secondly, we compare run-time performance. Finally, we showcase our model's ability to minimize the users' dependency on the grid, especially during peak demand, by highlighting aspects of a solution to one case.

#### 5.1 Model statistics

Because of the complicated mathematical structure of our model and the size of our cases, we reformulate model  $(\bar{\mathcal{R}})$  as  $(\hat{\mathcal{R}})$  using the methods described in Section 4. Table 2 shows the improvements in reformulation  $(\hat{\mathcal{R}})$  as a percent reduction in the number of variables and constraints. We concede that this reduction comes at the expense of a denser A matrix, owing to a more "compact" set of constraints, but the reduction in problem size more than offsets the increased density. That is, the approximately 40% reduction in both the number of variables and in the number of constraints in the reformulated model more than offsets the approximately same increase in the density of the constraint matrix when comparing solve times of the two models; this can be attributed to our use of simplex-based (versus interior point) methods. Furthermore, our reformulated model contains data that is much better scaled than that of the original model, where the decrease between the largest and smallest non-zero values is five (or more) orders of magnitude.

#### 5.2 Model performance

Model  $(\overline{\mathcal{R}})$  is implemented in Mosel (FICO 2021b) while model  $(\widehat{\mathcal{R}})$  is implemented in AMPL (Fourer et al. 2004); the difference in modeling language is the result of a legacy formulation versus a convenient instrument that enabled us to easily implement the enhancements we describe in Section 4, respectively. They are both solved using default algorithmic settings in Xpress V8.8.0 (FICO 2021a) on a Dell Power Edge R410 server with two Intel Xeon E5520s at 2.27 GHz 28GB RAM, and 1TB HDD. We display problem characteristics associated with each case, and provide the corresponding performance in terms of percent optimality gap achieved within a 10-minute time limit, which was deemed to be appropriate given that our model is embedded in a web-based tool. Table 3 shows that, in each case, model ( $\hat{\mathcal{R}}$ ) performs better than ( $\hat{\mathcal{R}}$ ) by achieving a tighter optimality gap. For each case in which the model finds a feasible solution, the gaps average more than 28% for model ( $\overline{\mathcal{R}}$ ), whereas the corresponding average gap is slightly greater than 2% for these same cases with model  $(\mathcal{R})$ . For Case 9, the only one that exercises every possible technology combined with both hot and cold thermal energy storage,  $(\overline{\mathcal{R}})$  cannot find a feasible solution within the time limit, while  $(\widehat{\mathcal{R}})$  finds a solution with a 2.65% optimality gap. Case 7 is another extreme that reaches only a 99.3% gap using  $(\overline{\mathcal{R}})$ , whereas  $(\widehat{\mathcal{R}})$  is able to reach a 1.11% optimality gap within the time limit. On average (excluding case 9), the overall gap improves by about 88% using formulation ( $\widehat{\mathcal{R}}$ ).

Because our formulation is an extension of model ( $\mathcal{R}$ ) in Ogunmodede et al. (2021), we also test those cases (i.e., without CHP) and find that our implementation outperforms the original with solution times that average between 30 and 60 seconds, amounting to a reduction of as much as two orders of magnitude (using the same software and hardware).

#### 5.3 Model dispatch strategy

We examine the solution determined by  $(\widehat{\mathcal{R}})$  for Case 9 which prescribes the systems shown in Table 4.

Figure 4 displays how the technologies are dispatched to meet hourly site requirements while reducing electrical power consumption from the utility relative to the business-as-usual scenario, which only employs the utility, the boiler, and the electric chiller to meet the electrical, heating, and cooling loads, respectively.

Typically, in an electrical demand graph (see Fig. 4a), there are five high-demand periods representing the afternoon and evening of each weekday. However, due to Christmas day occurring on Monday of the respective week, there are four. The dashed line highlights the business-as-usual scenario in which the hospital's cooling load is entirely met by the electric chiller. The optimized solution for Case 9 exhibits battery discharge, PV, and CHP—reducing the peak utility consumption to 212 kW to meet the majority of the electrical load. Figure 4b highlights the dispatch strategy of the CHP and the boiler systems. As part of peak-shaving, the absorption chiller is used instead of the electric chiller, and CHP is run at capacity. The heat provided by CHP cannot meet both the load consumed by the absorption chiller and the site heating load; therefore,

							<sup>2</sup> Gap (%)	
Case	<sup>1</sup> Technologies included	$\left \mathcal{U}\right $	$ \mathcal{N} $	$ \mathcal{D} $	Standby charges	Building type	$(\bar{\mathcal{R}})$	$(\widehat{\mathcal{R}})$
1	CHP, BOIL	1	1	30	No	Hospital	12.6	0.93
2	CHP, BOIL	1	1	30	No	Hospital	15.7	3.26
3	CHP, BOIL, TES	1	1	30	No	Hospital	17.6	4.12
4	CHP, BOIL, PV, BES	1	1	30	No	Hospital	15.8	4.32
5	CHP, BOIL, PV, BES	1	5	12	No	Hospital	35.1	1.32
6	CHP, BOIL, PV, BES	2	2	12	No	Hospital	46.6	0.46
7	CHP, BOIL, PV, BES	5	1	12	No	Hospital	99.3	1.11
8	CHP, BOIL, PV, BES	1	1	30	Yes	Hospital	8.70	0.71
9	CHP, BOIL, PV, EC, BES, TES, AC	1	1	30	No	Hospital	_	2.65
10	CHP, BOIL	1	1	30	No	Large office	11.2	0.59
11	CHP, BOIL	1	1	30	No	Large hotel	26.8	0.97
12	CHP, BOIL	1	1	30	No	Apartment	22.5	6.72

**Table 3** Results comparing solution quality obtained from model  $(\hat{\mathcal{R}})$  and model  $(\hat{\mathcal{R}})$  within a 10-min time limit, using AMPL and Mosel Xpress default settings

<sup>1</sup>BOIL: Boiler, EC: Electric Chiller, TES: Thermal Energy Storage, BES: Battery Electrical Storage, AC: Absorption Chiller, PV: Solar Photovoltaics.

<sup>2</sup>The gap is reported within a 10-minute time limit and is calculated as:  $\left(\frac{\text{Upper bound-Lower bound}}{\text{Upper bound}}\right) \cdot 100$ 

	Power	Ener	gy
Technology	[kW]	[kWh]	[gal]
CHP	789	_	_
PV	1400	-	-
Boiler	1696	-	-
Absorption chiller	512	-	-
Electric chiller	324	-	_
Battery energy storage	180	616	_
Chilled water thermal energy storage	_	_	19

**Table 4** Model  $(\widehat{\mathcal{R}})$ 's technology mix for Case 9

the boiler makes up the difference. Figure 4c demonstrates the dispatch strategies of the absorption chiller, electric chiller and the chilled water thermal storage system serving the cooling demand. The use of the absorption chiller reduces the dependence of the system on the existing electric chiller, thereby reducing the total electricity usage. The solution associated with model  $(\hat{\mathcal{R}})$  represents a 22% savings over business-as-usual.



(c) Case 9 Cooling Demand

Fig. 4 Dispatch summary for one week within Case 9, in which the technologies reduce peak electrical power consumption from the utility while meeting all hourly site loads

# 6 Conclusions

We examine a mixed-integer program that designs and dispatches renewable energy technologies and combined heat and power with a grid option. Instances of our mixed-integer program contain hundreds of thousands of variables and constraints, rendering an initial instantiation of our model intractable. To improve performance, we tailor data structures, efficiently handle data, and streamline the formulation through variable redefinition. These enhancements result in solutions within about 2% of optimality, on average, for the cases we test within a ten-minute time limit, rendering use of

the model appropriate for a web-based tool. Without the enhancements, instances generally remain at optimality gaps well above 10% within the same time limit.

Future work could entail implementing a decomposition procedure to further expedite solutions, and employing the model in international settings such as emerging markets of sub-Saharan Africa where opportunities for combined heat and power could enhance economic growth. Additionally, experimental work in thermal science might reveal that a more detailed combined heat and power representation might better reflect the operations of this technology. Finally, domestic implementation calls for myriad variations of the model and its output, such as considering alternate objective functions that would encompass resiliency and efficiency, and examining a diversity of solutions that would capture intangibles.

Acknowledgements This work was authored by the National Renewable Energy Laboratory, operated by Alliance for Sustainable Energy, LLC, for the US Department of Energy (DOE) under Contract No. DE-AC36-08GO28308. Funding was provided by the US Department of Energy Office of Energy Efficiency and Renewable Energy Federal Energy Management Program and the Advanced Manufacturing Office. The authors would like to thank Kate Anderson, Dylan Cutler, Bill Becker, Emma Elgqvist, Dan Olis, Ted Kwasnik, Nick Laws, Sakshi Mishra, Linda Parkhill, Kathleen Krah, Andrew Jeffery, Nick Muerdter, Rob Eger, and Andy Walker for help with building and maintaining the REopt Lite model. The authors would also like to thank Rachel Shepherd (Office of Energy Efficiency and Renewable Energy, DOE) and Bob Gemmer (Advanced Manufacturing Office, DOE) for funding this work. The views expressed in the article do not necessarily represent the views of the DOE or the US Government. The US Government retains an onexclusive, paid-up, irrevocable, worldwide license to publish or reproduce the published form of this work or allow others to do so, for US Government purposes.

# Appendix

We provide here additional notation not given (or used) in the body of the document, but that appears in the following constraints.

$\mathcal{K}$ Subdivisions of power rating $\mathcal{S}$ Power rating segments $\mathcal{V}$ Net metering regimesSubsets and indexed sets $\mathcal{M}^{lb}$ Look-back months considered for peak pricing $\mathcal{T}_u \subseteq \mathcal{T}$ Technologies that may access electrical energy sales pricing tier $u$ $\mathcal{U}^c \subseteq \mathcal{U}^s$ Electrical energy curtailment pricing tiers $\mathcal{U}^{nm} \subseteq \mathcal{U}^s$ Electrical energy sales pricing tiers used in net metering $\mathcal{U}^s \subseteq \mathcal{U}$ Electrical energy sales pricing tiers $\mathcal{Parameters for costs and their functional formsc^{amc}Utility annual minimum charge[\$]$	
S       Power rating segments         V       Net metering regimes         Subsets and indexed sets $\mathcal{M}^{lb}$ Look-back months considered for peak pricing $\mathcal{T}_u \subseteq \mathcal{T}$ Technologies that may access electrical energy sales pricing tier $u$ $\mathcal{U}^c \subseteq \mathcal{U}^s$ Electrical energy curtailment pricing tiers $\mathcal{U}^{nm} \subseteq \mathcal{U}^s$ Electrical energy sales pricing tiers used in net metering $\mathcal{U}^s \subseteq \mathcal{U}$ Electrical energy sales pricing tiers         Parameters for costs and their functional forms       [\$]	
$\mathcal{V}$ Net metering regimesSubsets and indexed sets $\mathcal{M}^{lb}$ Look-back months considered for peak pricing $\mathcal{T}_u \subseteq \mathcal{T}$ Technologies that may access electrical energy sales pricing tier $u$ $\mathcal{U}^c \subseteq \mathcal{U}^s$ Electrical energy curtailment pricing tiers $\mathcal{U}^{nm} \subseteq \mathcal{U}^s$ Electrical energy sales pricing tiers used in net metering $\mathcal{U}^s \subseteq \mathcal{U}$ Electrical energy sales pricing tiers $\mathcal{Parameters for costs and their functional formsc^{amc}Utility annual minimum charge[\$]$	
Subsets and indexed sets $\mathcal{M}^{lb}$ Look-back months considered for peak pricing $\mathcal{T}_u \subseteq \mathcal{T}$ Technologies that may access electrical energy sales pricing tier $u$ $\mathcal{U}^c \subseteq \mathcal{U}^s$ Electrical energy curtailment pricing tiers $\mathcal{U}^{nm} \subseteq \mathcal{U}^s$ Electrical energy sales pricing tiers used in net metering $\mathcal{U}^s \subseteq \mathcal{U}$ Electrical energy sales pricing tiers         Parameters for costs and their functional forms       [\$]	
$\mathcal{M}^{lb}$ Look-back months considered for peak pricing $\mathcal{T}_u \subseteq \mathcal{T}$ Technologies that may access electrical energy sales pricing tier $u$ $\mathcal{U}^c \subseteq \mathcal{U}^s$ Electrical energy curtailment pricing tiers $\mathcal{U}^{nm} \subseteq \mathcal{U}^s$ Electrical energy sales pricing tiers used in net metering $\mathcal{U}^s \subseteq \mathcal{U}$ Electrical energy sales pricing tiersParameters for costs and their functional forms $c^{amc}$ Utility annual minimum charge	
$\begin{array}{ll} \mathcal{T}_{u} \subseteq \mathcal{T} & \text{Technologies that may access electrical energy sales pricing tier } u \\ \mathcal{U}^{c} \subseteq \mathcal{U}^{s} & \text{Electrical energy curtailment pricing tiers} \\ \mathcal{U}^{nm} \subseteq \mathcal{U}^{s} & \text{Electrical energy sales pricing tiers used in net metering} \\ \mathcal{U}^{s} \subseteq \mathcal{U} & \text{Electrical energy sales pricing tiers} \\ \hline Parameters for costs and their functional forms \\ \hline c^{amc} & \text{Utility annual minimum charge} \\ \end{array} $	
$\begin{array}{ll} \mathcal{U}^{c} \subseteq \mathcal{U}^{s} & \text{Electrical energy curtailment pricing tiers} \\ \mathcal{U}^{nm} \subseteq \mathcal{U}^{s} & \text{Electrical energy sales pricing tiers used in net metering} \\ \mathcal{U}^{s} \subseteq \mathcal{U} & \text{Electrical energy sales pricing tiers} \\ \hline Parameters for costs and their functional forms \\ \hline c^{amc} & \text{Utility annual minimum charge} \\ \end{array} $	
$\begin{array}{ll} \mathcal{U}^{nm} \subseteq \mathcal{U}^{s} & \text{Electrical energy sales pricing tiers used in net metering} \\ \mathcal{U}^{s} \subseteq \mathcal{U} & \text{Electrical energy sales pricing tiers} \\ \hline Parameters for costs and their functional forms \\ \hline c^{amc} & \text{Utility annual minimum charge} \\ \end{array} $	
$\mathcal{U}^{s} \subseteq \mathcal{U}$ Electrical energy sales pricing tiersParameters for costs and their functional forms $c^{amc}$ Utility annual minimum charge[\$]	
Parameters for costs and their functional forms c <sup>amc</sup> Utility annual minimum charge [\$]	
c <sup>amc</sup> Utility annual minimum charge [\$]	
Power rating and fuel limit parameters	
$\underline{b}_{c}^{\sigma}$ Minimum power rating for technology class $c$ [kW]	
$\underline{b}_{tks}^{\sigma s}$ Minimum power rating for technology t, subdivision k, segment s [kW]	
$\bar{b}_{tks}^{\sigma s}$ Maximum power rating for technology t, subdivision k, segment s [kW]	
Binary variables	
$Z_{mn}^{\text{dmt}}$ 1 if tier <i>n</i> has allocated demand during month <i>m</i> ; 0 otherwise [unit]	ess]
$Z_{de}^{dt}$ 1 if tier <i>e</i> has allocated demand during time-of-use period <i>d</i> ; 0 otherwise [unit]	ess]
$Z_{v}^{\text{nmil}}$ 1 if generation is in net metering interconnect limit regime v; 0 otherwise [unit]	ess]
$Z_t^{p_1}$ 1 if production incentive is available for technology t; 0 otherwise [unit]	ess]
$Z_{mu}^{\text{ut}}$ 1 if demand tier <i>u</i> is active in month <i>m</i> ; 0 otherwise [unit]	ess]

The following constraints, when combined with those given in Section 3, form the monolith  $(\widehat{\mathcal{R}})$  that considers combined heat and power technologies in addition to the original renewable technologies. These constraints and their corresponding descriptions are taken directly from Ogunmodede et al. (2021) and are included here for completeness.

#### **Production incentives**

.

$$X_t^{\mathrm{pi}} \le \min\left\{\bar{\iota}_t \cdot Z_t^{\mathrm{pi}}, \sum_{h \in \mathcal{H}} \Delta \cdot i_t^{\mathrm{r}} \cdot f_t^{\mathrm{pi}} \cdot f_{th}^{\mathrm{p}} \cdot f_t^{\mathrm{li}} \cdot X_{th}^{\mathrm{rp}}\right\} \quad \forall t \in \mathcal{T}$$
(6a)

$$X_t^{\sigma} \le \bar{\imath}_t^{\sigma} + M \cdot (1 - Z_t^{\text{pi}}) \quad \forall t \in \mathcal{T}$$
(6b)

Constraint (6a) calculates total production incentives, if available, for each technology. Constraint (6b) sets an upper bound on the size of system that qualifies for production incentives, if production incentives are available.

#### Power rating

$$X_t^{\sigma} \le \bar{b}_t^{\sigma} \cdot \sum_{s \in \mathcal{S}_{tk}} Z_{tks}^{\sigma s} \quad \forall c \in \mathcal{C}, t \in \mathcal{T}_c, k \in \mathcal{K}_t$$
(7a)

$$\sum_{t \in \mathcal{T}_{c}, s \in \mathcal{S}_{tk}} Z_{tks}^{\sigma s} \le 1 \quad \forall c \in \mathcal{C}, k \in \mathcal{K}$$
(7b)

$$\sum_{t \in \mathcal{T}_c} X_t^{\sigma} \ge \underline{b}_c^{\sigma} \quad \forall c \in \mathcal{C}$$
(7c)

$$X_{th}^{\rm rp} = X_t^{\sigma} \quad \forall t \in \mathcal{T}^{\rm td}, h \in \mathcal{H}$$
(7d)

$$X_{th}^{\rm rp} \le f_{th}^{\rm ed} \cdot X_t^{\sigma} \quad \forall t \in \mathcal{T} \setminus \mathcal{T}^{\rm td}, h \in \mathcal{H}$$
(7e)

$$\underline{b}_{tks}^{\sigma_{s}} \cdot Z_{tks}^{\sigma_{s}} \le X_{tks}^{\sigma_{s}} \le \bar{b}_{tks}^{\sigma_{s}} \cdot Z_{tks}^{\sigma_{s}} \quad \forall t \in \mathcal{T}, k \in \mathcal{K}_{t}, s \in \mathcal{S}_{tk}$$
(7f)

$$\sum_{s \in \mathcal{S}_{tk}} X_{tks}^{\sigma s} = X_t^{\sigma} \quad \forall t \in \mathcal{T}, k \in \mathcal{K}_t$$
(7g)

Constraint (7a) permits nonzero power ratings only for the selected technology and corresponding subdivision in each class. Constraint (7b) allows at most one technology to be chosen for each subdivision in each class. Constraint (7c) limits the power rating to the minimum allowed for a technology class. Constraint (7d) prevents renewable technologies from turning down; rather, they must provide output at their nameplate capacity. Constraint (7e) limits rated production from all non-renewable technologies to be less than or equal to the product of the power rating and the derate factor for each time period. Constraint (7f) imposes both lower and upper limits on power rating of a technology, allocated to a subdivision in a segment, and constraint (7g) sums the segment sizes to the total for a given technology and subdivision.

#### Load balancing and grid sales

$$\sum_{t \in \mathcal{T}^{e}} (f_{th}^{p} \cdot f_{t}^{l} \cdot X_{th}^{rp}) + \sum_{b \in \mathcal{B}^{e}} X_{bh}^{dfs} + \sum_{u \in \mathcal{U}^{p}} X_{uh}^{g} = \sum_{t \in \mathcal{T}^{e}} (\sum_{b \in \mathcal{B}^{e}} X_{bth}^{pts} + \sum_{u \in \mathcal{U}_{t}^{s}} X_{tuh}^{ptg}) + \sum_{u \in \mathcal{U}^{sb}} X_{uh}^{stg} + X_{h}^{gts} + \sum_{t \in \mathcal{T}^{ec}} X_{th}^{tp} / \eta^{ec} + \delta_{h}^{d} \quad \forall h \in \mathcal{H}^{g}$$
(8a)

$$\sum_{t \in \mathcal{T}^{e}} (f_{th}^{p} \cdot f_{t}^{1} \cdot X_{th}^{rp}) + \sum_{b \in \mathcal{B}^{e}} X_{bh}^{dfs} = \sum_{b \in \mathcal{B}^{e}, t \in \mathcal{T}^{e}} \left( X_{bth}^{pts} + \sum_{u \in \mathcal{U}^{c}} X_{tuh}^{ptg} \right) + \sum_{u \in \mathcal{U}^{c}} X_{th}^{tp} / \eta^{ec} + \delta_{h}^{d} \quad \forall h \in \mathcal{H} \setminus \mathcal{H}^{g}$$
(8b)

 $t \in T^{ec}$ 

$$\sum_{u \in \mathcal{U}^{\mathsf{p}}} X_{uh}^{\mathsf{g}} \ge X_{h}^{\mathsf{gts}} \quad \forall h \in \mathcal{H}^{\mathsf{g}}$$
(8c)

$$\sum_{b \in \mathcal{B}^{\mathsf{e}}} X_{bh}^{\mathsf{dfs}} \ge \sum_{u \in \mathcal{U}^{\mathsf{sb}}} X_{uh}^{\mathsf{stg}} \quad \forall h \in \mathcal{H}^{\mathsf{g}}$$
(8d)

Springer

$$\Delta \cdot \sum_{h \in \mathcal{H}^{g}} \left( X_{uh}^{\text{stg}} + \sum_{t \in \mathcal{T}_{u}} X_{tuh}^{\text{ptg}} \right) \le \bar{\delta}_{u}^{\text{gs}} \quad \forall u \in \mathcal{U}^{\text{sb}} \cap \mathcal{U}^{\text{nm}}$$
(8e)

$$\Delta \cdot \sum_{h \in \mathcal{H}^{g}, t \in \mathcal{T}_{u}} X_{tuh}^{\text{ptg}} \le \bar{\delta}_{u}^{\text{gs}} \quad \forall u \in \mathcal{U}^{\text{nm}} \setminus \mathcal{U}^{\text{sb}}$$
(8f)

Constraint (8a) balances load by requiring that the sum of power (i) produced, (ii) discharged from storage, and (iii) purchased from the grid is equal to the sum of (i) the power charged to storage, (ii) the power sold to the grid from in-house production or storage, (iii) the power charged to storage directly from the grid, (iv) any additional power consumed by the electric chiller (where this is an additional term relative to the original model  $(\mathcal{R})$ , and (v) the electrical load on site. Constraint (8b) provides an analogous load-balancing requirement for hours in which the site is disconnected from the grid due to an outage (and contains the same additional term relative to the original model  $(\mathcal{R})$ ). Constraint (8c) restricts charging of storage from grid production to the grid power purchased for each hour. Similarly, constraint (8d) restricts the sales from the electrical storage system to its rate of discharge in each time period. Constraints (8e) and (8f) restrict the annual energy sold to the grid at net-metering rates; only one of these is implemented in each case according to user-specified options. While a collection of pre-specified technologies may contribute to net-metering rates in both cases, constraint (8e) allows storage to contribute to net-metering while constraint (8f) does not.

#### **Rate tariff constraints**

Net Metering

$$\sum_{\nu \in \mathcal{V}} Z_{\nu}^{\text{nmil}} = 1 \tag{9a}$$

$$\sum_{t \in \mathcal{I}_v} f_t^{\mathrm{d}} \cdot X_t^{\sigma} \le i_v^{\mathrm{n}} \cdot Z_v^{\mathrm{nmil}} \quad \forall v \in \mathcal{V}$$
(9b)

$$\Delta \cdot \sum_{h \in \mathcal{H}^{g}} \left( \sum_{u \in \mathcal{U}^{nm}, t \in \mathcal{T}_{u}} X_{tuh}^{ptg} + \sum_{u \in \mathcal{U}^{nm} \cap \mathcal{U}^{sb}} X_{uh}^{stg} \right) \le \Delta \cdot \sum_{u \in \mathcal{U}^{p}, h \in \mathcal{H}^{g}} X_{uh}^{g} \quad (9c)$$

Constraint (9a) limits the net metering to a single regime at a time. Constraint (9b) restricts the sum of the power rating of all technologies to be less than or equal to the net metering regime. Constraint (9c) ensures that energy sales at net-metering rates do not exceed the energy purchased from the grid.

Monthly Total Demand Charges

$$\Delta \cdot \sum_{h \in \mathcal{H}_m} X_{uh}^{g} \le \bar{\delta}_u^{\text{tu}} \cdot Z_{mu}^{\text{ut}} \quad \forall m \in \mathcal{M}, u \in \mathcal{U}^{p}$$
(10a)

$$Z_{mu}^{\text{ut}} \le Z_{m,u-1}^{\text{ut}} \quad \forall u \in \mathcal{U}^{\text{p}} : u \ge 2, m \in \mathcal{M}$$
(10b)

Deringer

$$\bar{\delta}_{u-1}^{\mathrm{tu}} \cdot Z_{mu}^{\mathrm{ut}} \le \Delta \cdot \sum_{h \in \mathcal{H}_m} X_{u-1,h}^{\mathrm{g}} \quad \forall u \in \mathcal{U}^{\mathrm{p}} : u \ge 2, m \in \mathcal{M}$$
(10c)

Constraint (10a) limits the quantity of electrical energy purchased from the grid in a given month from a specified pricing tier to the maximum available. Constraint (10b) forces pricing tiers to be charged in a specific order, and constraint (10c) forces one pricing tier's purchases to be at capacity if any charges are applied to the next tier.

Peak Power Demand Charges: Months

$$X_{mn}^{\mathrm{dn}} \le \bar{\delta}_n^{\mathrm{mt}} \cdot Z_{mn}^{\mathrm{dmt}} \quad \forall n \in \mathcal{N}, m \in \mathcal{M}$$
(11a)

$$Z_{mn}^{\text{dmt}} \le Z_{m,n-1}^{\text{dmt}} \quad \forall n \in \mathcal{N} : n \ge 2, m \in \mathcal{M}$$
(11b)

$$\bar{\delta}_{n-1}^{\text{mt}} \cdot Z_{mn}^{\text{dmt}} \le X_{m,n-1}^{\text{dm}} \quad \forall n \in \mathcal{N} : n \ge 2, m \in \mathcal{M}$$
(11c)

$$\sum_{n \in \mathcal{N}} X_{mn}^{\mathrm{dn}} \ge \sum_{u \in \mathcal{U}^{\mathrm{p}}} X_{uh}^{\mathrm{g}} \quad \forall m \in \mathcal{M}, h \in \mathcal{H}_{m}$$
(11d)

Constraint (11a) limits the energy demand allocated to each tier to no more than the maximum demand allowed. Constraint (11b) forces monthly demand tiers to become active in a prespecified order. Constraint (11c) forces demand to be met in one tier before the next demand tier. Constraint (11d) defines the peak demand to be greater than or equal to all of the demands across the time horizon, where an equality is actually induced by the sense of the objective function. A user-defined option precludes CHP technology production from reducing peak demand; if selected, constraint (11d) becomes:

$$\sum_{n \in \mathcal{N}} X_{mn}^{\mathrm{dn}} \geq \sum_{u \in \mathcal{U}^{\mathrm{p}}} X_{uh}^{\mathrm{g}} + \sum_{t \in \mathcal{T}^{\mathrm{CHP}}} \left( f_{th}^{\mathrm{p}} \cdot f_{t}^{\mathrm{l}} \cdot X_{th}^{\mathrm{rp}} - \sum_{b \in \mathcal{B}^{\mathrm{h}}} X_{bth}^{\mathrm{pts}} - \sum_{u \in \mathcal{U}_{t}^{\mathrm{s}}} X_{tuh}^{\mathrm{ptg}} \right)$$
$$\forall m \in \mathcal{M}, h \in \mathcal{H}_{m}.$$

Peak Power Demand Charges: Time-of-Use Demand and Ratchet Charges

$$X_{de}^{de} \le \bar{\delta}_e^{t} \cdot Z_{de}^{dt} \quad \forall e \in \mathcal{E}, d \in \mathcal{D}$$
(12a)

$$Z_{de}^{dt} \le Z_{d,e-1}^{dt} \quad \forall e \in \mathcal{E} : e \ge 2, d \in \mathcal{D}$$
(12b)

$$\bar{\delta}_{e-1}^{t} \cdot Z_{de}^{dt} \le X_{d,e-1}^{de} \quad \forall e \in \mathcal{E} : e \ge 2, d \in \mathcal{D}$$
(12c)

$$\sum_{e \in \mathcal{E}} X_{de}^{de} \ge \max\left\{\sum_{u \in \mathcal{U}^{p}} X_{uh}^{g}, \delta^{lp} \cdot X^{plb}\right\} \quad \forall d \in \mathcal{D}, h \in \mathcal{H}_{d}$$
(12d)

$$X^{\text{plb}} \ge \sum_{n \in \mathcal{N}} X_{mn}^{\text{dn}} \quad \forall m \in \mathcal{M}^{\text{lb}}$$
(12e)

Constraints (12a)-(12d) correspond to constraints (11a)-(11d), respectively, but pertain to a type of charge not related to monthly use, but rather to time of use within a month. These *ratchet charges* are implemented using constraints (12d). The charge applied for

each time-of-use period is a linearizable function of the greater of the peak electrical demand during that period (as given by the first term on the right-hand side of (12d)) and a fraction of the peak demand that occurs over a collection of months (known as *look-back months*) during the year (as given by the second term on the right-hand side of (12d)). Constraint (12e) ensures the peak demand over the set of look-back months is no lower than the peak demand for each look-back month. In this way, charges are based not only on use in a given month, but also on a fraction of use over the last several months, and becomes relevant when this latter use is high relative to current use. If CHP technologies are not allowed to reduce peak demand, constraint (12d) becomes:

$$\sum_{e \in \mathcal{E}} X_{de}^{de} \ge \sum_{u \in \mathcal{U}^{p}} X_{uh}^{g} + \sum_{t \in \mathcal{T}^{CHP}} \left( f_{th}^{p} \cdot f_{t}^{1} \cdot X_{th}^{rp} - \sum_{b \in \mathcal{B}^{h}} X_{bth}^{pts} - \sum_{u \in \mathcal{U}_{t}^{s}} X_{tuh}^{ptg} \right)$$
$$\forall d \in \mathcal{D}, h \in \mathcal{H}_{d}.$$

#### Minimum utility charge

$$X^{\mathrm{mc}} \geq c^{\mathrm{amc}} - \underbrace{\left(\Delta \cdot \sum_{u \in \mathcal{U}^{\mathrm{p}}, h \in \mathcal{H}^{\mathrm{g}}} c_{uh}^{\mathrm{g}} \cdot X_{uh}^{\mathrm{g}} + \sum_{\substack{d \in \mathcal{D}, e \in \mathcal{E} \\ \mathrm{Time-of-Use \ Demand \ Charges}}} c_{de}^{\mathrm{r}} \cdot X_{de}^{\mathrm{de}} \right)$$

$$+ \underbrace{\sum_{\substack{m \in \mathcal{M}, n \in \mathcal{N} \\ \mathrm{Monthly \ Demand \ Charges}}}_{\mathrm{Monthly \ Demand \ Charges}} - \underbrace{\Delta \cdot \left(\sum_{\substack{h \in \mathcal{H}^{\mathrm{g}} \\ u \in \mathcal{U}^{\mathrm{sb}}}} c_{uh}^{\mathrm{e}} \cdot X_{uh}^{\mathrm{stg}} + \sum_{\substack{t \in \mathcal{T}, u \in \mathcal{U}_{t}^{\mathrm{s}}}} c_{uh}^{\mathrm{e}} \cdot X_{tuh}^{\mathrm{ptg}}\right)\right)}_{\mathrm{Energy \ Export \ Payment}}$$

$$(13)$$

Constraint (13) enforces a minimum payment to the utility provider, which is a fixed

constant less charges incurred from grid energy, time-of-use demand and monthly demand payments, plus sales from exports to the grid.

#### Non-negativity

$$X^{\text{plb}}, X^{\text{mc}} \ge 0 \tag{14a}$$

$$X_t^{\sigma}, X_t^{\mathrm{pi}} \ge 0 \quad \forall t \in \mathcal{T}$$
(14b)

$$X_{tuh}^{\text{ptg}} \ge 0 \quad \forall u \in \mathcal{U}, t \in \mathcal{T}_u, h \in \mathcal{H}$$
(14c)

$$X_{uh}^{\text{stg}}, X_{uh}^{\text{g}} \ge 0 \quad \forall u \in \mathcal{U}, h \in \mathcal{H}$$
(14d)

$$X_{de}^{\text{de}} \ge 0 \quad \forall d \in \mathcal{D}, e \in \mathcal{E}$$
(14e)

$$X_{mn}^{\mathrm{dn}} \ge 0 \quad \forall m \in \mathcal{M}, n \in \mathcal{N}$$
(14f)

$$X_h^{\text{gts}} \ge 0 \quad h \in \mathcal{H} \tag{14g}$$

$$X_b^{\rm bkW}, X_b^{\rm bkWh} \ge 0 \quad b \in \mathcal{B} \tag{14h}$$

$$X_{tks}^{\sigma s} \ge 0 \quad \forall t \in \mathcal{T}, k \in \mathcal{K}, s \in \mathcal{S}_{tk}$$
(14i)

$$X_{bth}^{\text{pts}} \ge 0 \quad \forall b \in \mathcal{B}, t \in \mathcal{T}, h \in \mathcal{H}$$
(14j)

$$X_{bh}^{\text{se}}, X_{bh}^{\text{dfs}} \ge 0 \quad \forall b \in \mathcal{B}, h \in \mathcal{H}$$
(14k)

$$X_{th}^{\text{rp}}, X_{th}^{\text{f}}, X_{th}^{\text{fb}}, X_{th}^{\text{tpb}}, X_{th}^{\text{tp}}, X_{th}^{\text{ptw}} \ge 0 \quad \forall t \in \mathcal{T}, h \in \mathcal{H}$$
(141)

#### Integrality

$$Z_v^{\text{nmil}} \in \{0, 1\} \quad \forall v \in \mathcal{V} \tag{15a}$$

$$Z_{tks}^{\sigma_{s}} \in \{0, 1\} \quad \forall t \in \mathcal{T}, k \in \mathcal{K}, s \in \mathcal{S}_{tk}$$
(15b)

 $Z_t^{\text{pi}} \in \{0, 1\} \quad \forall t \in \mathcal{T}$  (15c)

$$Z_{th}^{\text{to}} \in \{0, 1\} \quad \forall t \in \mathcal{T}, h \in \mathcal{H}$$
(15d)

$$Z_{de}^{dt} \in \{0, 1\} \quad \forall d \in \mathcal{D}, e \in \mathcal{E}$$
(15e)

$$Z_{mn}^{\text{dmt}} \in \{0, 1\} \quad \forall m \in \mathcal{M}, n \in \mathcal{N}$$
(15f)

$$Z_{mu}^{\text{ut}} \in \{0, 1\} \quad \forall m \in \mathcal{M}, u \in \mathcal{U}$$
(15g)

Finally, constraints (14) ensure all of the variables in our formulation assume nonnegative values. In addition to non-negativity restrictions, constraints (15) establish the integrality of the appropriate variables.

# References

- Adam A, Fraga ES, Brett DJ (2015) Options for residential building services design using fuel cell based micro-CHP and the potential for heat integration. Appl Energy 138:685–694
- Anderson K, Olis D, Becker B, Parkhill L, Laws N, Li X, Mishra S, Kwasnik T, Jeffrey A, Elgqvist E, Krah K, Cutler D, Zolan A, Muerdter N, Eger R, Walker A, Hampel C, Tomberlin G (2021) REopt Lite user manual. Technical Report NREL/TP-7A40-79235, National Renewable Energy Laboratory, Golden, CO (United States)
- Anyenya GA, Braun RJ, Lee KJ, Sullivan NP, Newman AM (2018) Design and dispatch optimization of a solid-oxide fuel cell assembly for unconventional oil and gas production. Optim Eng 19(4):1037–1081
- Blackburn L, Young A, Rogers P, Hedengren J, Powell K (2019) Dynamic optimization of a district energy system with storage using a novel mixed-integer quadratic programming algorithm. Optim Eng 20(2):575–603
- Bracco S, Dentici G, Siri S (2016) DESOD: a mathematical programming tool to optimally design a distributed energy system. Energy 100:298–309

- Braslavsky JH, Wall JR, Reedman LJ (2015) Optimal distributed energy resources and the cost of reduced greenhouse gas emissions in a large retail shopping centre. Appl Energy 155:120–130
- Brown GG, Dell RF (2007) Formulating integer linear programs: a rogues' gallery. INFORMS Trans Educ 7(2):153–159
- Buoro D, Pinamonti P, Reini M (2014) Optimization of a distributed cogeneration system with solar district heating. Appl Energy 124:298–308
- Burer M, Tanaka K, Favrat D, Yamada K (2003) Multi-criteria optimization of a district cogeneration plant integrating a solid oxide fuel cell-gas turbine combined cycle, heat pumps and chillers. Energy 28:497–518
- Camm JD, Raturi AS, Tsubakitani S (1990) Cutting big M down to size. Interfaces 20(5):61-66
- Chiradeja P, Ramakumar R (2004) An approach to quantify the technical benefits of distributed generation. IEEE Trans Energy Convers 19(4):764–773
- Connolly D, Lund H, Mathiesen BV, Leahy M (2010) A review of computer tools for analysing the integration of renewable energy into various energy systems. Appl Energy 87(4):1059–1082
- Cutler DS, Olis DR, Elgqvist EM, Li X, Laws ND, DiOrio NA, Walker HA, Anderson KH (2017) REopt: a platform for energy system integration and optimization. https://doi.org/10.2172/1395453
- De Mel IA, Klymenko OV, Short M (2020) Balancing accuracy and complexity in optimisation models of distributed energy systems and microgrids: a review. arXiv e-prints pp arXiv–2008
- El-Khattam W, Salama MM (2004) Distributed generation technologies, definitions and benefits. Electric Power Syst Res 71(2):119–128
- FICO (2021a) FICO<sup>®</sup> Xpress Solver. https://www.fico.com/en/products/fico-xpress-solver
- FICO (2021b) Mosel language reference manual. https://www.fico.com/fico-xpress-optimization/docs/ latest/mosel\_lang/dhtml/GUID-D73D2F47-371D-3394-AC89-A299E0F7F291.html
- Fourer R, Gay DM, Kernighan BW (2004) Design principles and new developments in the AMPL modeling language. Modeling languages in mathematical optimization. Springer, New York, pp 105–135
- Fuentes-Cortés LF, Flores-Tlacuahuac A (2018) Integration of distributed generation technologies on sustainable buildings. Appl Energy 224:582–601
- Goodall G, Scioletti M, Zolan A, Suthar B, Newman A, Kohl P (2019) Optimal design and dispatch of a hybrid microgrid system capturing battery fade. Optim Eng 20(1):179–213
- Gopalakrishnan H, Kosanovic D (2014) Economic optimization of combined cycle district heating systems. Sustain Energy Technol Assess 7:91–100
- Gopalakrishnan H, Kosanovic D (2015) Operational planning of combined heat and power plants through genetic algorithms for mixed 0–1 nonlinear programming. Comput Oper Res 56:51–67
- Gumerman EZ, Bharvirkar RR, LaCommare KH, Marnay C (2003) Evaluation framework and tools for distributed energy resources. Technical report, Lawrence Berkeley National Lab. (LBNL), Berkeley, CA (United States)
- Hamilton WT, Husted MA, Newman AM, Braun RJ, Wagner MJ (2020) Dispatch optimization of concentrating solar power with utility-scale photovoltaics. Optim Eng 21(1):335–369
- Hamilton WT, Newman AM, Wagner MJ, Braun RJ (2020) Off-design performance of molten salt-driven Rankine cycles and its impact on the optimal dispatch of concentrating solar power systems. Energy Convers Manage 220:113025
- Hollermann DE, Goerigk M, Hoffrogge DF, Hennen M, Bardow A (2021) Flexible here-and-now decisions for two-stage multi-objective optimization: method and application to energy system design selection. Optim Eng 22:821–847
- Huster WR, Schweidtmann AM, Mitsos A (2019) Working fluid selection for organic rankine cycles via deterministic global optimization of design and operation. Optim Eng 21:517–536
- Karlsson K, Meibom P (2008) Optimal investment paths for future renewable based energy systems-using the optimisation model Balmorel. Int J Hydrogen Energy 33(7):1777–1787
- Kerr T (2008) Evaluating the benefits of greater global investment in combined heat and power. Int Energy Agency
- Klotz E, Newman AM (2013) Practical guidelines for solving difficult linear programs. Surv Oper Res Manag Sci 18(1–2):1–17
- Klotz E, Newman AM (2013) Practical guidelines for solving difficult mixed integer linear programs. Surv Oper Res Manag Sci 18(1–2):18–32
- Koivisto M, Gea-Bermúdez J, Sørensen P (2019) North Sea offshore grid development: combined optimisation of grid and generation investments towards 2050. IET Renew Power Gener 14(8):1259–1267

- Krug R, Mehrmann V, Schmidt M (2020) Nonlinear optimization of district heating networks. Optim Eng 22:783–819
- Mashayekh S, Stadler M, Cardoso G, Heleno M (2017) A mixed integer linear programming approach for optimal DER portfolio, sizing, and placement in multi-energy microgrids. Appl Energy 187:154–168
- Merkel E, McKenna R, Fichtner W (2015) Optimisation of the capacity and the dispatch of decentralised micro-CHP systems: a case study for the UK. Appl Energy 140:120–134
- Mishra S, Pohl J, Laws N, Cutler D, Kwasnik T, Becker W, Zolan A, Anderson K, Olis D, Elgqvist E (2021) Computational framework for behind-the-meter DER techno-economic modeling and optimization: REopt Lite. Energy Systems (2021):1–29
- Nasution EF, Shadiq J, Purnomo HS, Socaningrumline JF (2019) Bali energy planning: optimization of energy resources for electrical generation 2019–2028. In: Proceedings of the 2019 2nd international conference on high voltage engineering and power systems (ICHVEPS), IEEE, pp 085–090
- Ogunmodede O, Anderson K, Cutler D, Newman A (2021) Optimizing design and dispatch of a renewable energy system. Appl Energy 287:116527
- Oluleye G, Allison J, Kelly N, Hawkes AD (2018) An optimisation study on integrating and incentivising thermal energy storage (TES) in a dwelling energy system. Energies 11(5):1095
- Perera ATD, Nik VM, Mauree D, Scartezzini JL (2017) Electrical hubs: an effective way to integrate nondispatchable renewable energy sources with minimum impact to the grid. Appl Energy 190:232–248
- Pruitt KA, Braun RJ, Newman AM (2013a) Establishing conditions for the economic viability of fuel cell-based, combined heat and power distributed generation systems. Appl Energy 111:904–920
- Pruitt KA, Braun RJ, Newman AM (2013b) Evaluating shortfalls in mixed-integer programming approaches for the optimal design and dispatch of distributed generation systems. Appl Energy 102:386–398
- Pruitt KA, Leyffer S, Newman AM, Braun RJ (2014) A mixed-integer nonlinear program for the optimal design and dispatch of distributed generation systems. Optim Eng 15(1):167–197
- Ringkjøb HK, Haugan PM, Solbrekke IM (2018) A review of modelling tools for energy and electricity systems with large shares of variable renewables. Renew Sustain Energy Rev 96:440–459
- Rong A, Lahdelma R (2007) CO2 emissions trading planning in combined heat and power production via multi-period stochastic optimization. Eur J Oper Res 176(3):1874–1895
- Scioletti MS, Newman AM, Goodman JK, Zolan AJ, Leyffer S (2017) Optimal design and dispatch of a system of diesel generators, photovoltaics and batteries for remote locations. Optim Eng 18(3):755– 792
- Siddiqui A, Marnay C, Firestone R, Zhou N (2005) Distributed generation with heat recovery and storage. Technical Report LBNL-58630, Lawrence Berkeley National Laboratory
- Silvente J, Kopanos GM, Pistikopoulos EN, Espuña A (2015) A rolling horizon optimization framework for the simultaneous energy supply and demand planning in microgrids. Appl Energy 155:485–501
- Stadler M, Groissböck M, Cardoso G, Marnay C (2014) Optimizing distributed energy resources and building retrofits with the strategic DER-CAModel. Appl Energy 132:557–567
- Trick M (2005) Formulations and reformulations in integer programming. International conference on integration of artificial intelligence (AI) and operations research (OR) techniques in constraint programming. Springer, New York, pp 366–379
- Weber C, Marechal F, Favrat D, Kraines S (2006) Optimization of an SOFC-based decentralized polygeneration system for providing energy services in an office-building in Tokyo. Appl Therm Eng 26:1409–1419
- Wiese F, Bramstoft R, Koduvere H, Alonso AP, Balyk O, Kirkerud JG, Tveten ÅG, Bolkesjø TF, Münster M, Ravn H (2018) Balmorel open source energy system model. Energ Strat Rev 20:26–34
- Zakrzewski T (2017) Advancing design sizing and performance optimization methods for building integrated thermal and electrical energy generation systems. PhD thesis, Illinois Institute of Technology
- Zhao B, Zhang X, Li P, Wang K, Xue M, Wang C (2014) Optimal sizing, operating strategy and operational experience of a stand-alone microgrid on Dongfushan Island. Appl Energy 113:1656–1666

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.